

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Stiebale, Joel; Wößner, Nicole

Working Paper M&As, investment and financing constraints

DICE Discussion Paper, No. 257

Provided in Cooperation with: Düsseldorf Institute for Competition Economics (DICE), Heinrich Heine University Düsseldorf

Suggested Citation: Stiebale, Joel; Wößner, Nicole (2017) : M&As, investment and financing constraints, DICE Discussion Paper, No. 257, ISBN 978-3-86304-256-1, Heinrich Heine University Düsseldorf, Düsseldorf Institute for Competition Economics (DICE), Düsseldorf

This Version is available at: https://hdl.handle.net/10419/162780

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Düsseldorf Institute for Competition Economics

DISCUSSION PAPER

No 257

M&As, Investment and Financing Constraints

Joel Stiebale, Nicole Wößner

July 2017



d | u | p düsseldorf university press

IMPRINT

DICE DISCUSSION PAPER

Published by

düsseldorf university press (dup) on behalf of Heinrich-Heine-Universität Düsseldorf, Faculty of Economics, Düsseldorf Institute for Competition Economics (DICE), Universitätsstraße 1, 40225 Düsseldorf, Germany www.dice.hhu.de

Editor:

Prof. Dr. Hans-Theo Normann Düsseldorf Institute for Competition Economics (DICE) Phone: +49(0) 211-81-15125, e-mail: <u>normann@dice.hhu.de</u>

DICE DISCUSSION PAPER

All rights reserved. Düsseldorf, Germany, 2017

ISSN 2190-9938 (online) – ISBN 978-3-86304-256-1

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editor.

M&As, Investment and Financing Constraints

Joel Stiebale Nicole Wößner¹

July 2017

Abstract

We use a panel data set of European firms to analyse the effects of domestic and international M&As on target firms' investment, growth and financial constraints. Combining propensity score matching with a difference-in-differences estimator, our results indicate that upon acquisition, target firms obtain better access to external finance, are characterized by higher levels of tangible and intangible assets, and display lower dependence of investments and cash savings to the availability of internal funds. We also provide evidence that these effects are concentrated among acquisitions during the 2007-2009 financial crisis, relatively small target firms, and domestic rather than foreign acquisitions.

JEL codes: F61, F23, G01, G34, L25, D22, D24 Keywords: Mergers and Acquisitions, Financial Constraints, Investment, Firm Growth, Financial Crisis, Foreign Ownership

¹Joel Stiebale, Duesseldorf Institute for Competition Economics (DICE), Heinrich-Heine University Duesseldorf, stiebale@dice.hhu.de; Nicole Wößner, DICE, Heinrich-Heine University Duesseldorf, woessner@dice.hhu.de. We would like to thank Andrea Ciani, Anna Gumpert, Jens Suedekum and seminar participants in Duesseldorf, Goettingen and Munich for helpful comments and suggestions.

1 Introduction

Mergers and acquisitions (M&As) play an increasingly important role in the world economy with a global transaction value that exceeded \$4 trillion for the first time in 2015. While acquisitions can create value and spur growth of target firms, critics claim that more than 50% of previous M&A deals have failed.¹ A growing body of theoretical and empirical research has analysed how M&As affect prices, productivity, innovation, employment, and wages. Recent empirical contributions suggest that acquisitions can also affect target firms' financial constraints.² This is an important finding, as target firms' ability to exploit growth opportunities and undertake productivity-enhancing investments critically depends on the availability of finance.

This paper analyses the following research questions: (1) What is the impact of acquisitions on financial constraints, investment and growth of target firms? (2) Do results vary with the characteristics of target firms and types of acquisitions? (3) Are the effects different for acquisitions that took place during the recent financial crisis?

To answer these research questions, we analyse more than 700 M&As in which European firms were acquired between 2003-2012. Our balance sheet data allows us to construct various indicators which are related to investments and financing constraints, including cash savings, debt, capital stock and intangible assets. Further, we also provide evidence on the effects of M&As on other outcomes that are likely to depend on access to finance such as target firms' growth rates of employment, sales, and productivity.

There are a couple of earlier empirical studies that are closely related to our paper. Wang and Wang (2015) show that Chinese target firms' financial conditions measured as liquid assets and debt ratios improve for foreign relative to domestically acquired firms. Since the importance of financial factors and the selection of target firms might differ substantially between Chinese and European markets, these results do not necessarily apply to M&As in developed countries.³ Erel et al. (2015) use a similar database as ours, but an earlier time period, and find that European target firms display higher levels of investment and a lower degree of financial constraints, indicated, for instance, by lower dependence of investment on the availability of internal finance. While they control for some observable characteristics at the firm- and industry-level, their empirical strategy is based on the

¹See, for instance, http://www.wsj.com/articles/2015-becomes-the-biggest-m-a-year-ever-1449187101 and https://hbr.org/2016/05/so-many-ma-deals-fail-because-companies-overlook-this-simple-strategy, accessed June 23, 2017.

 $^{^{2}}$ Erel et al. (2015) and Wang and Wang (2015) relate M&As to proxies for financing constraints. Recent contributions that analyse the effects of M&As on innovation, prices, productivity and other outcome variables include Guadalupe et al. (2012); Ashenfelter et al. (2014); Braguinsky et al. (2015); Javorcik and Poelhekke (2017), to mention a few. See section two for a detailed discussion of related literature.

 $^{^{3}}$ Previous research has also shown that acquisitions by financial companies such as private equity firms can lead to lower financing constraints and induce higher investment and innovation (Amess et al., 2016; Boucly et al., 2011), but it is unclear whether these mechanism also apply to M&As in general.

assumption of exogenous selection of acquisition targets. In the case of endogenous selection, it is challenging to identify whether acquired firms are less financially constrained due to acquisitions per se or whether acquirers select firms with better financial performance.

In this paper, we try to overcome this problem by applying propensity score matching to construct an adequate control group of non-acquired firms with similar characteristics. We then compare changes in outcome variables around the time of acquisition events between acquired firms and the control group using a difference-in-differences (DiD) estimator. Further, we contribute to the literature by including the 2007-2009 financial crisis in our analysis and compare the effects of M&As across time periods with presumably different degrees of financial constraints. Compared to earlier studies, we also analyse a broader set of outcome variables and various heterogeneous effects across characteristics of acquiring firms and acquisition targets.

Our results indicate that, on average, acquisitions lead to higher growth of assets in target firms which is consistent with M&As relaxing liquidity constraints. In line with the common perception that financial constraints are particularly relevant for intangible investment, we find that this effect is most pronounced for the growth of intangible assets. Changes in acquirers' assets around the time of acquisitions indicate that this increase cannot be explained by income shifting in which intangibles are transferred from acquirer's to target's balance sheet.

Previous research shows that the degree of cash holdings is associated with financing constraints since managers use cash as an insurance towards future financial shocks (e.g., Opler et al., 1999). Our results indicate that target firms' cash holdings fall upon acquisition while debt ratios increase which is consistent with lower vulnerability to financial shocks and improved access to external finance. Changes in these variables are concentrated among target firms of relatively small size – which are arguably more likely to be financially constrained – and in acquisitions during the 2007-2009 financial crisis.

We also estimate a positive average effect of acquisitions on the growth of capital, sales, and employment. However, these effects are only significant for acquisitions that took place during a narrowly defined financial crisis period, which indicates the importance of financial factors for the effects of acquisitions on firm growth. Controlling for endogenous selection of acquisitions targets seems to be important in our sample, as acquirers, on average, tend to invest in relatively large, productive, and financially liquid target firms, but show a partly different selection profile during the financial crisis.

To measure financial constraints more directly, we also use the matched sample to estimate investment-cash flow and cash-cash flow sensitivities and allow for heterogeneous effects between acquired and non-acquired firms before and after acquisition. Following Fazzari et al. (1988) and Almeida and Campello (2007), these sensitivities increase with the cost premium for external finance and thus the degree of credit constraints. We find that both investment-cash flow and cash-cash flow sensitivities fall significantly after acquisitions which is, again, consistent with M&As alleviating financial constraints in target firms.

The international economics literature argues that the characteristics of acquiring and target firms can be quite different in cross-border acquisitions (see, for instance, Guadalupe et al., 2012; Nocke and Yeaple, 2007). To test whether these characteristics matter for investment and financial indicators in target firms, we perform a separate matching and DiD analysis for international and domestic M&As. The average effects of M&As in our sample are mainly driven by domestic transactions, while most changes in our outcome variables of interest are statistically insignificant for cross-border M&As. This result can be explained by a selection effect since target firms in international acquisitions are significantly larger and more productive compared to domestic acquisition targets.

The rest of this paper is organized as follows. Section 2 discusses the related theoretical and empirical literature, section 3 provides a description of the data and section 4 describes the empirical strategy. Results of the empirical analysis are presented in section 5 and section 6 concludes.

2 Related literature

Capital markets are characterized by significant agency problems and information asymmetries between management and shareholders and between a firm's management and its creditors (e.g., Jensen and Meckling, 1976; Stiglitz and Weiss, 1981; Myers and Majluf, 1984; Jensen, 1986). Due to these information asymmetries, suppliers of finance are confronted with an adverse selection problem leading to the rationing of finance and external sources of financing being more expensive than internal sources. Some profitable investment projects will thus not be undertaken due to financial constraints. M&As might alleviate target firms' liquidity constraints due to access to the acquirers' resources or relationships with financiers, higher collateral of the merged entity, or signaling to potential providers of finance (Erel et al., 2015; Boucly et al., 2011). We test this hypothesis in the empirical analysis by analyzing growth rates of target firms pre and post acquisition and investigate how investments and cash savings of target firms depend on the availability of internal finance.

Since asymmetric information problems are arguably more pronounced for investment in innovative assets than for tangible investment, and the collateral value of intangible assets is limited, financial constraints are especially relevant for the financing of research and development (R&D) and other types of intangible investments (Brown et al., 2012). Further, the riskiness of R&D makes debt financing particularly difficult to obtain, since in contrast to equity market investors, creditors do not benefit from upside returns (Brown et al., 2009; Hsu et al., 2014). If M&As decrease financing constraints, it is likely that we see a higher increase in the accumulation of intangible compared to tangible assets.

The extent to which firms are financially constrained is likely to vary across types of firms, countries, and time periods. For instance, large firms should experience little financing constraints due to relatively high collateral, stricter reporting requirements – which reduce information asymmetries between firms and financiers – and better access to equity markets. In contrast, low collateral and the difficulty to convey information to providers of finance make it more likely that small firms have to rely on internal financial resources and have limited access to bank loans (e.g., Behr et al., 2013; Carpenter and Petersen, 2002). If acquirers alleviate financial constraints in target firms, we would thus expect higher effects for acquisition targets that are initially relatively small. In our empirical analysis, we thus estimate heterogeneous effects according to the pre-acquisition size of target firms.

Our estimation sample covers the recent global financial crisis, a period where credit constraints were particularly important and had huge impact on economic outcomes (see, e.g. Chodorow-Reich, 2014). If target firms benefit from acquisition through lower financing constraints, we should therefore see larger effects for M&As during the crisis. We are not aware of a study that compares the effects of M&As in crisis- and non-crisis periods, especially with regard to the financial channel of acquisitions.

International M&As can be quite different from domestic ones. The foreign direct investment (FDI) literature argues that due to large sunk costs of entering a foreign market, only firms with superior productivity can operate abroad profitably (Helpman et al., 2004). Part of the knowledge underlying this superior productivity might be transferred to target firms (Markusen, 2002; Ekholm and Hakkala, 2007; Arkolakis et al., 2013) and lead to improved profitability and financial conditions, implying lower financial constraints.⁴ Recently, the literature has argued that foreign acquisitions might be driven by liquidity of multinational firms (Alquist et al., 2014) and benefit target firms due to lower financing costs (e.g., Wang and Wang, 2015). It is likely that the effects of cross-border acquisitions depend on the timing of acquisitions as well. For instance, financial shocks and changes

⁴The productivity advantage of multinationals has, for instance, been related to management practices (e.g. Bloom and van Reenen, 2010) and differences in innovation and knowledge (e.g., Guadalupe et al., 2012). Besides knowledge transfer, foreign acquisitions might also benefit acquisition targets due to access to new markets (Guadalupe et al., 2012) or complementary assets of the acquiring firm (e.g., Nocke and Yeaple, 2007, 2008). Several empirical studies have documented significant performance gains in the form of productivity improvements in target firms after international M&As (e.g. Arnold and Javorcik, 2009; Chen, 2011; Guadalupe et al., 2012) while other scholars have argued that the effects of cross-border M&As are not that different from other ownership changes (e.g. Gugler et al., 2003; Fons-Rosen et al., 2013; Wang and Wang, 2015). There is a large literature on the effects of M&As on efficiency-related outcomes which either analyses domestic transactions or does not explicitly distinguish between domestic and international M&As. This literature indicates that domestic acquisitions can lead to productivity gains as well (e.g., Maksimovic and Phillips, 2001; David, 2013; Braguinsky et al., 2015).

in local demand during the 2007-2009 financial crisis differed across types of firms and countries.

A number of empirical studies show that foreign owned firms are less likely to be financially constrained than domestic firms (Bridges and Guariglia, 2008; Guariglia and Mateut, 2010; Harrison and McMillan, 2003), are less affected by local currency devaluations that increase debt (Desai et al., 2008) and display higher growth rates compared to domestically owned firms during the years of the financial crisis (Alfaro and Chen, 2012). However, these studies analyse cross-sectional differences in foreign ownership and therefore cannot isolate the causal effects of foreign acquisitions. Alquist et al. (2014) provide evidence that multinationals particularly target domestic firms in financially dependent industries and in countries with low levels of financial development. However, the authors only analyse financial factors as a determinant of FDI and foreign acquisitions, not whether financial constraints in target firms are indeed reduced as a result of being acquired. There is not much evidence on the effects of foreign acquisitions on target firms' liquidity or financial constraints. A notable exception is Wang and Wang (2015) who analyse the effects of international acquisitions in China before the years of the financial crisis.

The degree to which different types of M&As can reduce financial constraints is likely to depend on the type of firms being acquired. While some theories suggest that acquirers tend to select firms of high productivity and size (e.g., Guadalupe et al., 2012), which are typically not very likely to be financially constrained, other scholars argue that there are incentives to invest in underperforming target firms (e.g., Neary, 2007) or that the selection profile depends on industry characteristics such as the type of capabilities that determine productivity (Nocke and Yeaple, 2007). Whether target firms are likely to face financing constraints before acquisitions, and domestic and international M&As thus have the potential to reduce these constraints, is therefore ultimately an empirical question.

3 Data

Data and sample selection We link financial data on European firms from the Amadeus database to data on domestic and international M&As between 2003 and 2012 from the Zephyr database, both provided by Bureau van Dijk. The Amadeus database contains financial information on public and private firms for 43 countries in Europe, including standardized annual accounts. We use information from unconsolidated accounts to separately identify economic activity in target firms, and match our financial variables to information from Zephyr using a common firm identifier. The Zephyr database contains M&A, IPO, private equity and venture capital deals, and provides information on various characteristics of the deal, e.g. date, deal value, deal type, stake, target and

buyer firms. By combining these two datasets, we are able to analyse financial information for target firms before and after an acquisition. In addition, we use data on firms which are not involved in M&As in the considered period to construct a control group.

The data are cleaned in the following way. First, we drop observations with implausible values, like negative employment, fixed assets, leverage, and cash holdings. Second, to deal with extreme outliers, we delete the lower and upper 0.5% quantile of each variable. In addition, we drop very small firms with a median value of operating revenue smaller than half a million euros or less than 5 employees based on all available observations per firm. Following Erel et al. (2015), we also address the concern that some firms' assets cannot be identified correctly after an acquisition if acquirers transfer some of their assets to the target firm. We therefore exclude firms with implausible large changes (changes of more than 100%) in the number of employees from one year to another from the analysis. This procedure is also applied to non-acquired firms, as large variations in employment might indicate an unreported merger. Finally, financial variables are deflated using data from the European Central Bank and Eurostat.⁵

The analysis is restricted to firms in the manufacturing sector (NACE Revision 2, 2-digit industry codes 10-33). In the main specification, we focus on completed majority acquisitions where the stake controlled rises from below to at least 50%. After restricting the sample to targets with information on all necessary variables for our empirical analysis, the final M&A sample consists of 736 deals between 2003 and 2012. Table 1 provides an overview of some deal characteristics. We refer to a cross-border deal if acquiring and target firm are located in different countries, whereas domestic deals cover M&As within the same country. The financial crisis period is defined as the period between the second half of 2007 and the end of 2009 (see, e.g., Flannery et al., 2013).⁶ Almost one quarter of deals took place during the recent financial crisis. On average, 41% of M&As are cross-border, though this share is increasing over time from 32% in the pre-crisis to 47% in the post-crisis period.⁷

Construction of variables We exploit information from firms' balance sheet and profit and loss accounts to construct our outcome and control variables. Data on sales, labour and capital stock (measured as tangible fixed assets) provide information on a firm's growth path, size, and capital intensity. Financial conditions and liquidity are captured by cash and cash equivalents, working capital (current assets less current liabilities), cash flow, and leverage which we measure relative to total assets. The average wage (costs of employees divided by number of employees)

⁵We use the *Gross domestic product at market price* deflator for EU-28 countries.

 $^{^{6}}$ We also distinguish between this rather broad and a more narrow crisis period in some regressions.

⁷The distribution of target firms across countries is depicted in Table A.1 in the Appendix.

approximates the average skill level in the firm, and the share of intangible fixed assets in fixed assets is used as a proxy for R&D. To obtain a measure of total factor productivity (TFP), we apply the generalized method of moments (GMM) estimation strategy suggested by Wooldridge (2009), controlling for unobserved productivity shocks using investments (Olley and Pakes, 1996). We estimate separate production functions per 2-digit NACE industry by relating operating revenue to capital, employment and material costs. We define firm-level TFP as the residual from this regressions which is thus estimated as the deviation of productivity from the industry-specific mean. Finally, we use information on the firm's age and legal form.⁸

4 Econometric strategy

The aim of our study is to identify the causal impact of M&As on target firms' financing constraints, investment and growth. For this purpose, we combine a DiD estimator with propensity score matching.

The effect of acquisition can be formalized by the average treatment effect on the treated (ATET)

$$\beta_{ATET} = E[y_{t+s}^1 | MA_t = 1] - E[y_{t+s}^0 | MA_t = 1], \tag{1}$$

where y_{t+s}^1 describes the observed outcome *s* periods after acquisition, and y_{t+s}^0 the hypothetical outcome in the absence of acquisition. MA_t is a binary variable equal to one if a firm is acquired in period *t* and zero otherwise. The second term $E[y_{t+s}^0|MA_t = 1]$ constitutes the counterfactual situation, i.e. the firm's outcome had it not been the target of a deal. We employ a matching procedure to obtain an estimate for this unobserved outcome by constructing a comparison group of firms which are not acquired, but observationally similar to target firms. Specifically, we use propensity score matching as proposed by Rosenbaum and Rubin (1983), and exploit balance sheet data to estimate the probability of acquisition. The average outcome of the comparison group is then used to identify the mean counterfactual outcome of target firms in the absence of acquisition. To illustrate, Equation 1 can be rewritten as follows:

$$\beta_{ATET} = E[y_{t+s}^1 | MA_t = 1, X_{t-1}] - E[y_{t+s}^0 | MA_t = 0, X_{t-1}] -(E[y_{t+s}^0 | MA_t = 1, X_{t-1}] - E[y_{t+s}^0 | MA_t = 0, X_{t-1}]).$$
(2)

The first term describes the difference in observed outcomes between acquired and non-acquired firms. The second term represents a comparison between the hypothetical outcome of acquired

⁸For variable definitions see Table A.2 in the Appendix.

firms had they not been acquired and the observed outcome of non-acquired firms. The latter is the bias we aim to minimize by selecting non-acquired firms which are as similar as possible to target firms with respect to pre-acquisition characteristics X_{t-1} .

We further combine the propensity score matching with a DiD estimator (e.g. Blundell and Costa Dias, 2000). The DiD estimator compares the targets' outcome in the period before acquisition with the outcome *s* periods afterwards, controlling for the difference in outcomes of matched non-acquired firms:

$$\beta_{DiD} = E[y_{t+s}^1 - y_{t-1}^1 | MA_t = 1, X_{t-1}] - E[y_{t+s}^0 - y_{t-1}^0 | MA_t = 0, X_{t-1}].$$
(3)

This procedure has the advantage that the assumption of selection on observables is relaxed by allowing the selection into acquisition to be correlated with time-invariant unobservable characteristics.

The estimation strategy consists of the following steps. First, we predict the probability that a firm is acquired in period t based on firm characteristics in t - 1 using a Probit model. The sample includes both acquired firms as well as firms which are not involved in M&As in the whole period under study. By means of the estimated probability, the so-called *propensity score*, we apply one-to-one nearest neighbor matching without replacement, i.e. each acquired firm is matched to one firm in the comparison group of non-acquired firms such that the difference in propensity scores is minimized.⁹ Using this matched data set, the DiD procedure involves simple ordinary least squares (OLS) regressions of the change in outcomes on the dummy for acquisition:

$$\Delta y_{it+s} = \alpha + \beta_{DiD} M A_{it} + \epsilon_{it}. \tag{4}$$

Heterogeneous effects of acquisition are calculated by testing for significance of $\hat{\beta}_{2,DiD}$ using the following specification:

$$\Delta y_{it+s} = \alpha_0 + \beta_{1,DiD} M A_{it} + \beta_{2,DiD} (M A_{it} \times small_i) + \beta_{3,DiD} small_i + \eta_{it}, \tag{5}$$

exemplified on the basis of the target firm's pre-acquisition size (proxied by a binary variable *small*). We examine effects on cash ratio, leverage ratio, capital stock, intangible assets ratio, sales and employment (all ratios are divided by the firm's total assets). To account for possible heteroscedasticity, we compute robust standard errors.¹⁰

Similar to Erel et al. (2015), we consider the change in cash holdings after acquisition. The idea is

⁹The choice of matching with or without replacement can be seen as a tradeoff between bias and variance. Since our sample of potential firms in the comparison group is large, we decide to perform matching without replacement.

 $^{^{10}}$ There is no need to cluster standard errors by firms since equations 4 and 5 are estimated separately by time period and therefore do not include repeated firm observations.

a precautionary motive in the presence of capital market imperfections, as described by Keynes (1936, p. 196): "To provide for contingenties requiring sudden expenditure and for unforeseen opportunities for advantageous purchases [...]", and taken up by e.g. Kim et al. (1998) and Opler et al. (1999). If the management maximizes shareholders' wealth, cash holdings are set such that their marginal benefit equals their marginal cost: holding cash is costly since it lowers the rate of return, but it is beneficial as it can be used to finance investments if external sources of financing are not available or simply too costly. Hence, as "there is no necessity to hold idle cash to bridge over intervals if it can be obtained without difficulty at the moment when it is actually required" (Keynes, 1936, p. 196), the amount of cash held by the firm should be higher the more it expects to face financing constraints in the future.¹¹ Consequently, if M&As reduce target firms' financing constraints, the cash ratio is expected to decrease after acquisition. Similarily, if firms obtain better access to capital markets, debt may be used as a substitute for holding liquid assets such as cash holdings.¹² Hence, in the presence of investment opportunities and financing constraints pre-acquisition, the leverage ratio should rise after a deal.¹³

We further test for increases in investment by analyzing the change in tangible and intangible fixed assets. If investments can be financed more easily as a result of improved access to finance, we expect assets to increase after a deal. This effect might be most pronounced for intangible assets, which are particularly prone to asymmetric information problems. Moreover, we analyse additional firm outcomes, namely labour and sales. Higher liquidity may reduce growth constraints, allowing target firms to hire new employees and generate higher output.

To obtain an alternative indicator of financing constraints, we estimate investment-cash flow and cash-cash flow sensitivities. Using the intuition that external finance is more costly than internal finance, a seminal paper by Fazzari et al. (1988) studies the relation between investment and internal finance in the presence of financing constraints. They argue that financially constrained firms, which rely on internal sources to finance investments, should display a positive sensitivity of investment to cash flow (as a proxy for internal funds). In contrast, investment decisions of unconstrained firms are not expected to depend on cash flow. The approach of Fazzari et al. (1988) is controversial and has been criticized extensively (e.g., Kaplan and Zingales, 1997; Erickson and Whited, 2000; Gomes, 2001; Cleary et al., 2007). The criticism involves that investment-cash flow sensitivities do

¹¹Empirical work such as Opler et al. (1999) show a negative relation between access to the capital market and the amount of cash held by the firm. More recently, Hadlock and Pierce (2010) also find that firms with more cash are more likely to be financially constrained.

 $^{^{12}}$ Kim et al. (1998) test this explanation and provide evidence that the leverage ratio is negatively related to liquid assets.

 $^{^{13}}$ We argue that increased leverage after acquisition is consistent with relaxed credit constraints as firms are able to rely more on external financial funds, in line with e.g. Bellone et al. (2010) and Boucly et al. (2011). In contrast, Wang and Wang (2015) interpret a decrease in the leverage ratio and an increase in the liquidity ratio as a reduction in financing constraints.

not increase monotonically with the level of financing constraints. In addition, critics argue that a positive cash flow coefficient may partly capture the correlation between cash flow and investment opportunities. However, this indicator has been applied in several recent contributions which find positive cash flow sensitivities for constrained firms even in cases where mismeasurement of investment opportunities is a minor concern (see, for instance, Almeida and Campello, 2007; Bond and Söderbom, 2013). In addition, it has been argued that even in the presence of measurement error, *differences* in investment-cash flow sensitivities are at least a useful measure of *differences* in financing constraints across different groups of firms (see also Erel et al., 2015). Hence, if acquirers facilitate access to external finance, we expect that target firms' investment depends to a lesser extent on cash flow.

To analyse how the cash flow sensitivity of investment changes after an acquisition, we estimate:

$$I_{it} = \gamma_0 + \gamma_1 CF_{it} + \gamma_2 after_{it} + \gamma_3 (after_{it} \times CF_{it}) + \gamma_4 X_{it} + d_t + \alpha_i + u_{it}, \tag{6}$$

where I_{it} measures investment in capital (calculated as the change in tangible fixed assets plus depreciation) of firm *i* at year *t*, *CF* denotes cash flow, and *after* is a binary variable which takes on value one after acquisition. *I* and *CF* are scaled by the beginning-of-year capital stock. *X* contains control variables, for instance, sales growth as a proxy for investment opportunities and year dummies (d_t). We are primarily interested in the change of the cash flow sensitivity of investment after acquisition: if firms' financing constraints decrease after a deal, we expect γ_3 the coefficient on the interaction term between *after* and *CF* to be negative and statistically significant. Taking into account the criticism that the sensitivity may not monotonically increase with the level of financing constraints, the results would be most convincing if the overall sensitivity was positive and significant in the absence of an acquisition (indicated by a positive coefficient γ_1), but small and insignificant after acquisition ($\gamma_1 + \gamma_3 \approx 0$). To deal with potential unobserved heterogeneity, we apply a fixed effects (FE) estimator, and use the results of the propensity score matching, i.e. the sample includes target firms and their matched controls. Standard errors are clustered at the firm level.

A related measure of financing constraints is the sensitivity of cash to cash flow (Almeida et al., 2004). The idea is that a financially constrained firm will increase its cash holdings due to an increase in cash flow to finance investments today and in the future, while an unconstrained firm's cash holdings do not systematically vary with cash flow.¹⁴ Since cash is a financial variable (rather than a variable capturing firm growth such as investment), the authors argue that this measure avoids some problems associated with the investment-cash flow sensitivity, for example that a positive cash

 $^{^{14}}$ Other authors, e.g. Khurana et al. (2006) and Hadlock and Pierce (2010), also provide empirical evidence that the cash flow sensitivity of cash is related to a firm's financing constraints.

flow coefficient simply captures the correlation between cash flow and investment opportunities.¹⁵ However, they point out that the cash-cash flow sensitivity of constrained firms does not necessarily increase monotonically with the degree of financing constraints, since the degree depends on the borrowing capacity and on the size of the firms' cash flows relative to their investment opportunities (Almeida et al., 2004, p. 1785). Similar to Equation 6, we use the following equation to estimate cash-cash flow sensitivities:

$$\triangle Cash_{it} = \delta_0 + \delta_1 CF_{it} + \delta_2 after_{it} + \delta_3 (after_{it} \times CF_{it}) + \delta_4 X_{it} + d_t + \alpha_i + w_{it}, \tag{7}$$

where $\triangle Cash$ is the change in cash holdings between two consecutive years. $\triangle Cash$ and CF are scaled by beginning-of-year total assets. We use the same estimation technique and sample as for the cash flow sensitivity of investment. If financing constraints decrease after acquisition, δ_3 is expected to be negative. Again, the most clear-cut result would be a positive and significant cash flow coefficient in the absence of an acquisition (δ_1), and a small and insignificant cash-cash flow sensitivity post-acquisition ($\delta_1 + \delta_3$).

5 Results

5.1 Effect of M&As on financial variables, investment, and firm growth Baseline specification

Results of the Probit estimation for the probability that a firm is acquired in a given period are shown in Table 2. In addition to firm characteristics, we include industry dummies at the NACE 2-digit level, year dummies, and country dummies in the estimation. Large firms with high liquidity, and firms with a high capital intensity are more likely to be acquired. In addition, public limited companies seem to be more likely to become the target of a deal. The negative coefficient for the change in sales indicates that target firms experience lower sales growth before acquisition. Higher average wages and a higher share of intangible assets also increase the probability of a deal. This is in line with the idea of *cherry-picking*, i.e. only the best performing firms within an industry are selected for acquisition (Guadalupe et al., 2012). The negative coefficient for TFP does not point in the same direction. However, as Table 3 shows, unconditional on other variables, target firms are significantly more productive than non-acquired ones. A reason for the negative coefficient for TFP

¹⁵However, their approach has been critized by Riddick and Whited (2009). They show in a dynamic framework that cash holdings are in fact negatively related to cash flow when accounting for measurement error in *Tobin's q*. In addition, they argue that the amount of cash savings is not only related to a firm's financing constraints, but also (and to a greater extent) to its income uncertainty.

might be that this variable is highly correlated with other included firm characteristics such as firm size, the average skill level, and the importance of intangible assets.

After performing the matching procedure, we check the balancing condition, i.e. whether conditional on the propensity score, selection into acquisition is independent of observable firm characteristics.¹⁶ In Table 3, we report t-tests for the equality of means between treated and non-acquired (control) firms. Before matching, we observe significant differences in the firm characteristics used in the Probit estimation: acquired firms are on average larger, more capital-intensive, more innovative and more productive than non-acquired firms. Hence, there is selection of firms into acquisition, which justifies our matching approach. After matching, however, the hypothesis of equal means cannot be rejected at conventional significance levels for all variables. In addition, the mean propensity scores do not significantly differ after matching, i.e. the average probability of being acquired is very similar in both groups.

The results of DiD estimations are displayed in Table 4. They are consistent with the view that M&As reduce financing constraints and foster growth in target firms. Compared to non-acquired firms, target firms experience on average a significant decline in the cash ratio. Considering the fact that the average cash ratio of target firms is 8.7% in the year before acquisition, a decline of 1.2percentage points in the second year after a deal implies a reduction of around 14%. The estimates are similar to Erel et al. (2015), who find that cash holdings over total assets are reduced by 1.4 to 1.7 percentage points. The decrease in the cash ratio comes along with a moderate increase in the leverage ratio. This effect is driven by a significant rise in long-term debt, suggesting that firms obtain better access to external finance after a deal.¹⁷ Since we use unconsolidated balance sheet information, i.e. financial statements of target firms themselves (in contrast to consolidated information which includes the parent company and its other subsidiaries), the increase in debt is unlikely to be due to borrowed funds in the context of the deal financing which would appear in the acquirers' accounts (see also Boucly et al., 2011, p. 435). With regard to investment, acquired firms increase their intangible assets ratio by 0.5 to 1.2 percentage points. This corresponds to a rise in intangible assets from around 17% in the year of the deal to 40% two years afterwards when taking into account that the mean intangible assets ratio is about 3% in the pre-acquisition year.¹⁸ In addition, we find a positive, albeit a smaller, impact on the capital stock which amounts to approximately 8.7% in the second year after a deal. The findings are qualitatively similar but

¹⁶Note that the common support condition is fulfilled for all acquired firms in the sample.

 $^{^{17}}$ See Table A.3 in the Appendix where we separately analyse the change in long-term debt and current liabilities, the two components of leverage.

 $^{^{18}}$ As a robustness check, we perform the DiD estimation for the log of intangible assets as an outcome variable (see Table A.3 in the Appendix). Since around 22% of acquired firms have zero intangible assets in the pre-acquisition year, the log of intangible assets is calculated as ln(Intangible Assets + 1). The estimates are highly significant and similar in magnitude.

smaller than in Erel et al. (2015) who estimate an increase in the mean firm's gross investment by 23% to 31%. While they do not distinguish between tangible and intangible fixed assets, our results indicate that the rise in investment is mainly driven by intangible assets. Moreover, we analyse changes in labour and sales, and find positive growth effects two years after acquisition.

One might be concerned that the large increase in intangible assets is related to transfer pricing and accompanied by a decrease in acquirers' intangible assets after acquisition. In Table A.4 in the Appendix, we therefore compare mean intangible assets of acquirers in the year before a deal with respective figures up to two years afterwards.¹⁹ Our results indicate that acquirers' intangibles do not decrease upon acquisition. This result holds both for the log of intangible assets and for the ratio of intangible assets to total assets, whereat for the former we even identify a statistically significant increase. Further, as we discuss below, most of the effects are concentrated in domestic deals, hence dispelling the concern that transfer pricing drives the results.

If target firms' financial constraints are indeed decreasing after acquisition, we should see greater effects for firms which are a priori more likely to be financially constrained and suffer from underinvestment. One proxy for the cost of external funds in the literature is firm size (e.g. Hennessy and Whited, 2007; Hadlock and Pierce, 2010; Erel et al., 2015), with smaller firms presumably facing higher costs due to higher information asymmetries and lower collateral. We calculate heterogeneous effects of acquisition with regard to firm size by interacting the treatment indicator with a binary variable *small* (Equation 5). A firm is treated as small if its pre-acquisition employment is below or equal the bottom tercile of acquired and matched control firms (48 employees). Table 5 illustrates that the effect of acquisition is especially pronounced for small firms. In particular, the decrease in the cash ratio and the increase in the capital stock are high in magnitude and statistically significant for this group of firms only. We also find a stronger rise in employment for small firms. However, with regard to investments in intangible assets, we do not find large differences related to firm's size.

Cross-border versus domestic acquisitions

The effects of M&As on target firms' financing constraints are likely to differ by the acquirers' origin. Foreign-acquired firms may benefit from greater access to (foreign) capital markets, which is likely to lower their costs of external sources of financing. This is especially important if acquiring firms active in international acquisitions are those which are highly productive (e.g. Helpman et al., 2004) and financially liquid (e.g. Alquist et al., 2014). However, the potential to decrease financing constraints after acquisition depends crucially on the degree target firms are constrained before a deal.

 $^{^{19}}$ Unfortunately, we only have data on intangible assets for a limited number of acquirers: the figures in Table A.4 are based on the 27% of deals for which we have information on acquirers' intangible assets in all three years.

We distinguish cross-border and domestic acquisitions, and perform the propensity score matching and DiD estimation separately for both deal types, considering the fact that the selection profile might differ.²⁰ The results of the Probit regressions are shown in Table A.5 in the Appendix. As for the whole sample of target firms, large firms in terms of employment, and those with high average wages and a high share of intangible assets are more likely to be selected into both domestic and foreign acquisitions. Targets of international acquisitions are more capital-intensive and productive, while domestic-acquired firms are more liquid than non-acquired ones. DiD estimates are displayed in Table 6.²¹ The results indicate that the average effects of acquisition and therefore the evidence for a reduction in financing constraints mainly stem from target firms in domestic deals: their cash ratio (leverage ratio) decreases (increases) significantly after acquisitions, there is some evidence of positive growth effects with regard to sales, pointing to another channel firms benefit from acquisition, namely the access to new markets.²²

One explanation for these findings is that domestic acquisitions per se lead to a larger decrease in firms' financing constraints. Similar to Wang and Wang (2015), we therefore directly compare the effects of cross-border and domestic M&As, and perform another propensity score matching using domestic-acquired firms as a control group.²³ Since the matching is aimed at finding control firms with similar pre-acquisition characteristics, we are able to assess directly if the effects found in Table 6 are due to the deal type or to targets' characteristics. While the leverage ratio significantly decreases for firms in cross-border relative to firms in domestic deals, we find neither significant differences in the cash ratio nor in assets accumulation (see Table A.9 in the Appendix). However, the effect for leverage disappears when additionally controlling for pre-acquisition age and cash flow in the DID estimation, the variables which are not completely balanced between foreign-acquired and domestically acquired firms after matching.²⁴ Hence, the deal type does not seem sufficient to explain the differences in outcomes between international and domestic acquisitions.

An alternative explanation is related to the selection of target firms into cross-border and do-

 $^{^{20}}$ For about 3% of acquired firms (=20 deals), we have no information on whether the deal is cross-border or domestic. These observations are excluded from this analysis.

²¹Again, we apply one-to-one nearest neighbor matching without replacement using non-acquired firms as a control group. The balancing condition is tested in Table A.6 in the Appendix.

 $^{^{22}}$ We also analysed if effects for cross-border deals differ by the acquirers' country or region but did not find a systematic pattern of heterogeneous effects.

 $^{^{23}}$ As the number of potential control firms (i.e. domestic-acquired firms) is only somewhat larger than the number of treated firms (i.e. foreign-acquired firms), we allow for propensity score matching with replacement. In addition, the common support condition is imposed, i.e. foreign-acquired firms which are off common support are not included (corresponds to 30 deals in our sample). Tables A.7 and A.8 in the Appendix display the results of the Probit estimation and the propensity score matching.

²⁴More precisely, we perform the DID estimation in Equation 4, while additionally controlling for the log of age and the cash flow ratio in period t-1, as well as their respective interaction with the treatment indicator MA (results available upon request).

mestic M&As. Summary statistics in Table A.8 in the Appendix show that, in the pre-acquisition year, target firms in domestic deals are significantly smaller than those in cross-border deals. The former are therefore more likely to be financially constrained before acquisition which increases the potential for reducing such constraints. As discussed before, the effect of acquisitions are particularly pronounced for small firms (which are overrepresented in domestic deals). Therefore, the different findings for domestic acquisition might be due to selection rather than the deal type per se.

Crisis versus non-crisis acquisitions

Our sample of M&As covers the recent global financial crisis, a period where credit constraints were particularly important and had huge impact on economic outcomes (see, e.g. Chodorow-Reich, 2014). If target firms benefit from acquisition through lower financing constraints, we should therefore see larger effects for M&As during the crisis.

To analyse effects of M&As during the crisis, we consider both a broad and a narrow crisis period: the broad period includes M&As between the second half of 2007 and the end of 2009, while the narrow period starts with the Lehman bankruptcy in September 2008. We perform the matching and DiD estimation separately for the three groups regarding their time of acquisition (narrow crisis, broad crisis, non-crisis period) since the motivation of parent firms to engage in an acquisition is likely to be different during an economic crisis.²⁵ When comparing target firms' characteristics across time periods, we see that firms differ, primarily in their TFP: those acquired during the crisis (narrow period) are equally productive than the industry average, while those involved in deals before and after the crisis are substantially more productive (about 15%) than the other firms in their industry.

Table 7 displays the results of DiD estimations for various outcomes. With the exception of the increase in intangible assets, the effects virtually disappear for M&As outside the financial crisis. For deals during the crisis, especially for the period after the Lehman bankruptcy, however, we observe a decrease in cash and a strong increase in the capital stock of about 23% two years after acquisition. In addition, compared to non-acquired firms, target firms experience a significant increase in sales and employment when taken over during the crisis. Hence, most of our effects are concentrated among M&As in crisis years, emphasizing the financial channel of acquisition which benefits target firms when financing constraints are presumably most severe.

When examining more closely the impact of M&As during the crisis, our results indicate that it is especially firms in technology-intensive industries which benefit from acquisition. In Table

 $^{^{25}}$ Again, we perform a one-to-one nearest neighbor matching without replacement. Firms not involved in M&As are used as a control group. The Probit estimations and balancing tests are given in Tables A.10, A.11 and A.12 in the Appendix.

A.13 in the Appendix, we show that the overall average effect of acquisition on the capital stock stems from target firms in high-tech industries.²⁶ By contrast, non-acquired firms in these industries significantly decrease their capital stock during the crisis. A similar pattern is found for the change in cash holdings, but this effect is not statistically significant. As firms in high-tech industries are likely to be particularly dependent on external finance, the findings for acquisitions during the financial crisis reinforce the idea that M&As may decrease liquidity constraints in target firms.²⁷

5.2 Effect of M&As on other outcomes

Cash flow sensitivities

To provide further evidence that M&As reduce target firms' financing constraints, we present estimation results for cash flow sensitivities of investment and of cash holdings. Again, the idea is to measure how investment and cash holdings respond to changes in internal funds. According to Fazzari et al. (1988) and Almeida et al. (2004), a financially constrained firm will increase its investment respectively cash holdings due to an increase in cash flow, while an unconstrained firm's investment (respectively cash holdings) should not systematically vary with cash flow. We are interested in how these sensitivities change after acquisition. Estimates for the cash flow sensitivity of investment (Panel A) and of cash (Panel B) are shown in Table 8. In the first column, we present fixed effects regressions using all acquired and matched control firms. The coefficient on cash flow is positive and significant in both equations, indicating that firms are indeed financially constrained.

The negative estimates for the interaction term between cash flow and the dummy *after* are consistent with the view that firms' financing constraints decrease after a deal. However, the absolute value of cash flow sensitivities after acquisitions (the sum of the coefficients for CF and $CF \times after$) are still significantly different from zero (see F-tests in Table 8). Our interpretation of results, therefore, has to be treated with caution when taking into account the criticism that the relationship between cash flow sensitivities and the level of financing constraints may not be monotonous. In the second and third column, we add additional control variables, namely sales growth as a proxy for investment opportunities, and the log of total assets to control for firms' size. The results remain similar. In the fourth column, we drop the year of the deal for acquired firms to make sure that

²⁶The classification is based on the aggregation of the manufacturing industry according to technological intensity by the European Commission (Eurostat), see http://ec.europa.eu/eurostat/cache/metadata/Annexes/htec_esms_ an3.pdf, accessed Jan 24, 2017. We distinguish high-technology (high tech and medium-high tech) and low-technology (low-tech and medium-low-tech) industries. To estimate heterogeneous effects by technological intensity, we exploit Equation 5 using acquired firms during the narrow crisis period and their matched controls, and replacing *small* with an indicator variable *hightech*.

 $^{^{27}}$ We do not observe heterogeneous effects by technological intensity for the sample of acquired firms over the whole sample period (see Table A.14 in the Appendix). One reason might be that high-tech firms are less financially constrained during non-crisis periods.

the findings are not solely driven by contemporaneous effects. While the interaction term between cash flow and *after* is no longer significant for the cash flow sensitivity of investment, the decrease in the cash-cash flow sensitivity after acquisition is still significant and high in magnitude. In the last column, we add a binary variable *Deal* (=1 for acquired firms, =0 for non-acquired firms), and interact it with cash flow to control for permanent differences in sensitivities that affect firms' behaviour both pre- and post-acquisition. The results are robust. All in all, the estimates in Table 8 support the previous findings that financing constraints decrease after M&As.

Total factor productivity

Previous empirical evidence shows that domestic and international M&As sometimes lead to productivity improvements in target firms (e.g. Arnold and Javorcik, 2009; Guadalupe et al., 2012; Wang and Wang, 2015). If these gains lead to higher profitability, acquired firms' improved financial conditions may simply be an indirect effect that stems from higher productivity post-acquisition. To examine whether this channel explains our previous results, we perform the DiD estimation with TFP as an outcome variable. In addition to the whole sample of target firms, productivity effects are estimated separately for cross-border and domestic acquisitions, and for firms acquired in crisis and non-crisis periods.²⁸ Moreover, we split firms based on their productivity in the pre-acquisition year, treating them as *low productive* if their TFP is below or equal the industry average (corresponds to 29% of acquired firms). To estimate heterogeneous effects of acquisition by initial productivity level, we exploit Equation 5 and replace the indicator of firm size *small* with a binary variable *lowprod* characterizing low productive firms. The results are illustrated in Table 9. In contrast to previous work, we do not find a significant impact of M&As on target firms' productivity on average. These findings do not change when considering different deal types, or acquisitions during and outside the recent financial crisis. Initially low productive firms, however, experience a increase in TFP of about 5% in the first and second year after acquisition. This result is consistent with productivity effects of M&As that are concentrated in low productive targets which presumably have most to learn from the superior productivity and knowledge of parent firms. All in all, our estimated productivity effects indicate that productivity improvements occur for some firms but are unlikely to be the main explanation for the relationship between M&As and financial indicators.

 $^{^{28}}$ To estimate separate productivity effects for different deal types and time periods, we draw on the results of the separate matching procedures (see Chapter 5.1).

5.3 Robustness checks

We perform several robustness checks which we apply to the baseline DiD specification. These are briefly summarized in this section and documented in the Appendix (Tables A.15 to A.19).

First, the robustness of the results towards the matching procedure is analysed. We initially stick to one-to-one nearest neighbor matching based on the propensity score, but conduct matching with replacement. In addition, regarding the propensity score estimation, we use a Logit instead of a Probit regression, and include some key variables two years before the deal to control for pre-acquisition trends.²⁹ We also experimented with other matching algorithms, and implement caliper matching which avoids bad matches by imposing a maximum tolerance level for the propensity score distance, radius matching which uses all non-acquired firms within a caliper as control firms, and two variants of Mahalanobis matching as well as kernel matching. With regard to Mahalanobis matching, we perform a one-to-one nearest neighbor matching on the propensity score while putting additional weight on the firm's year and country (first specification), and on the firm's year and industry (second specification), in order to rule out the concern that acquired firms are compared to control firms in different countries or industries. The main findings are not affected by this variation in matching algorithms. However, the effects on the capital stock, labour and sales are smaller and become insignificant in some specifications. Nevertheless, the general result that acquisitions spur growth in target firms still holds, whether in terms of capital, employment, or sales.

Second, our approach to combine propensity score matching with a DiD estimator is only valid if the Stable Unit Treatment Value Assumption (SUTVA) is fulfilled, i.e. if outcomes of matched non-acquired firms are not affected by acquisitions. Since we compare similar firms in terms of observable characteristics within industry, year and country, this assumption could be violated – particularly if a control firm is located in the same region within a country as the corresponding acquired firm. We therefore exclude those pairs and reestimate the DiD regression.³⁰ Since the results remain almost unchanged, we argue that a violation of this assumption is a minor concern in our case.

Third, our matching strategy and DiD estimates are compared to results on an unmatched sample. We perform OLS regressions similar to Equation 4; however, we do not rely on the matched data set, but use target firms and all non-acquired firms (not only matched ones). In the first OLS specification, the change in outcomes is regressed on the dummy MA, lagged firm characteristics X_{t-1} used to estimate the propensity score, and time, year and country dummies. In another

 $^{^{29}\}mathrm{The}$ key variables include cash ratio, leverage ratio, capital stock, employment, and intangible fixed assets over fixed assets.

 $^{^{30}}$ This applies to 4 pairs of acquired and matched control firms. Unfortunately, for 276 firms (196 acquired firms, 171 control firms) we do not have information about their region.

specification, we regress the change in outcomes on the dummy MA, and apply propensity score reweighting, i.e. each acquired firm is weighted by 1 and each firm in the comparison group by $\hat{p}/(1-\hat{p})$ with \hat{p} being the estimated propensity score. Standard errors are clustered at the firmlevel. Our previous findings for cash ratio, leverage ratio, intangible assets ratio, sales and labour are confirmed. As in some other robustness checks, the change in the capital stock becomes smaller in magnitude and statistically insignificant.

6 Conclusion

Agency problems and information asymmetries on capital markets can lead to the rationing of finance. Firms which are financially constrained may not be able to exploit investment opportunities because their cost premium for external finance is too high. M&As may alleviate target firms' liquidity constraints since, for instance, being part of a larger organization may ease access to bank loans due to higher collateral and signaling to providers of finance. Previous literature has primarily looked at the effects of M&As on efficiency-related outcomes, but evidence on the financial channel of acquisitions is still scarce.

This paper provides evidence on the impact of domestic and cross-border M&As on financing constraints, investment, and growth in acquired firms. We use a panel data set of European target firms exploiting balance sheet information before and after acquisition, and combine propensity score matching with a DiD estimator. The results are consistent with the view that M&As reduce financing constraints and foster growth in target firms. Our findings indicate that, on average, acquisition targets hold less cash but increase leverage, suggesting that they obtain better access to capital markets and are less in need of liquid assets for precautionary reasons. In addition, we find higher growth of assets after a deal, and increases in employment and sales. The results vary across firm types, deal types, and the time of M&As. Some of our estimated effects are most pronounced for relatively small targets which are likely to be particularly constrained. In addition, there are stronger responses for those acquired during the recent financial crisis, highlighting the financial channel of acquisition which benefits target firms by decreasing liquidity constraints when they are most severe. Interestingly, the effects are concentrated among domestic acquisitions, while most changes in outcomes are statistically insignificant for firms acquired in international M&As. However, this finding seems to be due to firms' pre-acquisition characteristics rather than the deal type per se.

When assessing the potential benefits of acquisitions from an economic policy point of view, financial factors such as the alleviation of liquidity constraints should be taken into account. This channel is important, since availability of finance is crucial for growth and value creation in acquisition targets. Our results indicate that an active market for corporate control may be particularly beneficial during time periods in which some firms face unusually severe credit constraints such as during the recent financial crisis.

For future research, it might be interesting to analyse the channels that lead to alleviation of financing constraints upon acquisition in more detail. This would create new insights on the motives of firms to engage in M&As and would therefore enrich the discussion on potential benefits of acquisitions. Further, it would be interesting to analyse a sample of target firms that is not limited to European countries and construct alternative measures of financial constraints.

References

- Alfaro, Laura and Maggie Xiaoyang Chen, "Surviving the Global Financial Crisis: Foreign Ownership and Establishment Performance," American Economic Journal: Economic Policy, 2012, 4 (3), 30–55.
- Almeida, Heitor and Murillo Campello, "Financial Constraints, Asset Tangibility, and Corporate Investment," *Review of Financial Studies*, 2007, 20 (5), 1429–1460.
- _, _, and Michael S. Weisbach, "The Cash Flow Sensitivity of Cash," Journal of Finance, 2004, 59 (4), 1777–1804.
- Alquist, Ron, Rahul Mukherjee, and Linda L. Tesar, "Liquidity-Driven FDI," Working Paper No. 17, Graduate Institute of International and Development Studies 2014.
- Amess, Kevin, Joel Stiebale, and Mike Wright, "The impact of private equity on firms patenting activity," *European Economic Review*, 2016, *86*, 147–160.
- Arkolakis, Costas, Natalia Ramondo, Andrés Rodríguez-Clare, and Stephen Yeaple, "Innovation and Production in the Global Economy," Working Paper No. 18972, National Bureau of Economic Research 2013.
- Arnold, Jens Matthias and Beata S. Javorcik, "Gifted kids or pushy parents? Foreign direct investment and plant productivity in Indonesia," *Journal of International Economics*, 2009, 79 (1), 42–53.
- Ashenfelter, Orley C., Daniel Hosken, and Matthew C. Weinberg, "Did Robert Bork Understate the Competitive Impact of Mergers? Evidence from Consummated Mergers," *Journal* of Law and Economics, 2014, 57 (S3), S67–S100.
- Behr, Patrick, Lars Norden, and Felix Noth, "Financial constraints of private firms and bank lending behavior," *Journal of Banking & Finance*, 2013, 37 (9), 3472–3485.
- Bellone, Flora, Patrick Musso, Lionel Nesta, and Stefano Schiavo, "Financial Constraints and Firm Export Behaviour," World Economy, 2010, 33 (3), 347–373.
- Bloom, Nicholas and John van Reenen, "Why Do Management Practices Differ across Firms and Countries?," *Journal of Economic Perspectives*, 2010, 24 (1), 203–224.
- Blundell, Richard and Monica Costa Dias, "Evaluation Methods for Non-Experimental Data," Fiscal studies, 2000, 21 (4), 427–468.

- Bond, Stephen R. and Måns Söderbom, "Conditional Investment-Cash Flow Sensitivities and Financing Constraints," *Journal of the European Economic Association*, 2013, 11 (1), 112–136.
- Boucly, Quentin, David Sraer, and David Thesmar, "Growth LBOs," Journal of Financial Economics, 2011, 102 (2), 432–453.
- Braguinsky, Serguey, Atsushi Ohyama, Tetsuji Okazaki, and Chad Syverson, "Acquisitions, Productivity, and Profitability: Evidence from the Japanese Cotton Spinning Industry," *American Economic Review*, 2015, 105 (7), 2086–2119.
- Bridges, Sarah and Alessandra Guariglia, "Financial Constraints, Global Engagement, and Firm Survival in the United Kingdom: Evidence from Micro Data," Scottish Journal of Political Economy, 2008, 55 (4), 444–464.
- Brown, James R., Gustav Martinsson, and Bruce C. Petersen, "Do financing constraints matter for R&D?," *European Economic Review*, 2012, 56 (8), 1512–1529.
- -, Steven M. Fazzari, and Bruce C. Petersen, "Financing Innovation and Growth: Cash Flow, External Equity, and the 1990s R&D Boom," *Journal of Finance*, 2009, 64 (1), 151–185.
- Carpenter, Robert E. and Bruce C. Petersen, "Is the Growth of Small Firms Constrained by Internal Finance?," *Review of Economics and statistics*, 2002, 84 (2), 298–309.
- Chen, Wenjie, "The effect of investor origin on firm performance: Domestic and foreign direct investment in the United States," *Journal of International Economics*, 2011, 83 (2), 219–228.
- Chodorow-Reich, Gabriel, "The Employment Effects of Credit Market Disruptions: Firm-level Evidence from the 2008–9 Financial Crisis," *Quarterly Journal of Economics*, 2014, 129 (1), 1–59.
- Cleary, Sean, Paul Povel, and Michael Raith, "The U-Shaped Investment Curve: Theory and Evidence," Journal of Financial and Quantitative Analysis, 2007, 42 (1), 1–39.
- David, Joel, "The Aggregate Implications of Mergers and Acquisitions," Research Paper No. 14.02, Center for Applied Financial Economics 2013.
- Desai, Mihir A., C. Fritz Foley, and Kristin J. Forbes, "Financial Constraints and Growth: Multinational and Local Firm Responses to Currency Depreciations," *Review of Financial Studies*, 2008, 21 (6), 2857–2888.
- Ekholm, Karolina and Katariina Hakkala, "Location of R&D and High-Tech Production by Vertically Integrated Multinationals," *Economic Journal*, 2007, 117 (518), 512–543.

- Erel, Isil, Yeejin Jang, and Michael S. Weisbach, "Do Acquisitions Relieve Target Firms" Financial Constraints?," Journal of Finance, 2015, 70 (1), 289–328.
- Erickson, Timothy and Toni M. Whited, "Measurement Error and the Relationship between Investment and q," *Journal of Political Economy*, 2000, 108 (5), 1027–1057.
- Fazzari, Steven M., R. Glenn Hubbard, and Bruce C. Petersen, "Financing Constraints and Corporate Investment," *Brookings papers on economic activity*, 1988, 19 (1), 141–206.
- Flannery, Mark J., Simon H. Kwan, and Mahendrarajah Nimalendran, "The 2007–2009 financial crisis and bank opaqueness," *Journal of Financial Intermediation*, 2013, 22 (1), 55–84.
- Fons-Rosen, Christian, Sebnem Kalemli-Ozcan, Bent E. Sørensen, Carolina Villegas-Sanchez, and Vadym Volosovych, "Quantifying Productivity Gains from Foreign Investment," Working Paper No. 18920, National Bureau of Economic Research 2013.
- Gomes, Joao F., "Financing Investment," American Economic Review, 2001, 91 (5), 1263–1285.
- Guadalupe, Maria, Olga Kuzmina, and Catherine Thomas, "Innovation and Foreign Ownership," American Economic Review, 2012, 102 (7), 3594–3627.
- Guariglia, Alessandra and Simona Mateut, "Inventory investment, global engagement, and financial constraints in the UK: Evidence from micro data," *Journal of Macroeconomics*, 2010, 32 (1), 239–250.
- Gugler, Klaus, Dennis C. Mueller, B. Burcin Yurtoglu, and Christine Zulehner, "The effects of mergers: an international comparison," *International Journal of Industrial Organization*, 2003, 21 (5), 625–653.
- Hadlock, Charles J. and Joshua R. Pierce, "New Evidence on Measuring Financial Constraints: Moving Beyond the KZ Index," *Review of Financial Studies*, 2010, 23 (5), 1909–1940.
- Harrison, Ann E. and Margaret S. McMillan, "Does direct foreign investment affect domestic credit constraints?," *Journal of International Economics*, 2003, 61 (1), 73–100.
- Helpman, Elhanan, Marc J. Melitz, and Stephen R. Yeaple, "Export versus FDI with Heterogeneous Firms," American Economic Review, 2004, 94 (1), 300–316.
- Hennessy, Christopher A. and Toni M. Whited, "How Costly Is External Financing? Evidence from a Structural Estimation," *Journal of Finance*, 2007, 62 (4), 1705–1745.

- Hsu, Po-Hsuan, Xuan Tian, and Yan Xu, "Financial development and innovation: Crosscountry evidence," *Journal of Financial Economics*, 2014, *112* (1), 116–135.
- Javorcik, Beata Smarzynska and Steven Poelhekke, "Former Foreign Affiliates: Cast Out and Outperformed?," Journal of the European Economic Association, 2017, 15 (3), 501–539.
- Jensen, Michael C., "Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers," Corporate Finance, and Takeovers. American Economic Review, 1986, 76 (2), 323–329.
- and William H. Meckling, "Theory of the firm: Managerial behavior, agency costs and ownership structure," Journal of Financial Economics, 1976, 3 (4), 305–360.
- Kaplan, Steven N. and Luigi Zingales, "Do Investment-Cash Flow Sensitivities Provide Useful Measures of Financing Constraints?," *Quarterly Journal of Economics*, 1997, 112 (1), 169–215.
- **Keynes, John M.**, *The General Theory of Employment, Interest, and Money*, London: Macmillan, 1936.
- Khurana, Inder K., Xiumin Martin, and Raynolde Pereira, "Financial Development and the Cash Flow Sensitivity of Cash," *Journal of Financial and Quantitative Analysis*, 2006, 41 (4), 787–807.
- Kim, Chang-Soo, David C. Mauer, and Ann E. Sherman, "The Determinants of Corporate Liquidity: Theory and Evidence," *Journal of Financial and Quantitative Analysis*, 1998, 33 (3), 335–359.
- Maksimovic, Vojislav and Gordon Phillips, "The Market for Corporate Assets: Who Engages in Mergers and Asset Sales and Are There Efficiency Gains?," *Journal of Finance*, 2001, 56 (6), 2019–2065.
- Markusen, James R., Multinational Firms and the Theory of International Trade, The MIT Press, 2002.
- Myers, Stewart C. and Nicholas S. Majluf, "Corporate financing and investment decisions when firms have information that investors do not have," *Journal of Financial Economics*, 1984, 13 (2), 187–221.
- Neary, Peter J., "Cross-Border Mergers as Instruments of Comparative Advantage," Review of Economic Studies, 2007, 74 (4), 1229–1257.

- Nocke, Volker and Stephen Yeaple, "Cross-border mergers and acquisitions vs. greenfield foreign direct investment: The role of firm heterogeneity," *Journal of International Economics*, 2007, 72 (2), 336–365.
- Nocke, Volker. and Stephen Yeaple, "An Assignment Theory of Foreign Direct Investment," *Review of Economic Studies*, 2008, 75 (2), 529–557.
- Olley, G. Steven and Ariel Pakes, "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, 1996, 64 (6), 1263–1297.
- **Opler, Tim, Lee Pinkowitz, René Stulz, and Rohan Williamson**, "The determinants and implications of corporate cash holdings," *Journal of Financial Economics*, 1999, 52 (1), 3–46.
- Riddick, Leigh A. and Toni M. Whited, "The Corporate Propensity to Save," Journal of Finance, 2009, 64 (4), 1729–1766.
- Rosenbaum, Paul R. and Donald B. Rubin, "The central role of the propensity score in observational studies for causal effects," *Biometrika*, 1983, 70 (1), 41–55.
- Stiglitz, Joseph E. and Andrew Weiss, "Credit Rationing in Markets with Imperfect Information," American Economic Review, 1981, 71 (3), 393–410.
- Wang, Jian and Xiao Wang, "Benefits of foreign ownership: Evidence from foreign direct investment in China," *Journal of International Economics*, 2015, 97 (2), 325–338.
- Wooldridge, Jeffrey M., "On estimating firm-level production functions using proxy variables to control for unobservables," *Economics Letters*, 2009, 104 (3), 112–114.

Tables

	Absolute	Share $(\%)$	Cross-border $(\%)$
Before the crisis	248	34	32
In the crisis	178	24	43
After the crisis	310	42	47
Total	736	100	41

Table 1: M&A deals between 2003 and 2012.

NOTES: For 3% of all deals it is not known if the deal is cross-border or domestic. Crisis: 07/2007-12/2009.

 Table 2: Probit regression. Prediction of M&A deals.

Cash ratio	0.0730	Leverage ratio	0.1618^{*}
	(0.1304)		(0.0846)
Working capital ratio	0.2329***	$\ln(Wage)$	0.2278^{***}
	(0.0861)		(0.0516)
$\ln(\text{Capital})$	0.0421^{***}	Public	0.0645^{*}
	(0.0148)		(0.0333)
$\ln(\text{Labour})$	0.4671^{***}	$\ln(Age)$	-0.0170
	(0.0714)		(0.0185)
$\ln(\text{Labour})^2$	-0.0280***	Intangible assets share	0.5379^{***}
	(0.0077)		(0.0853)
$\Delta \ln(\text{Sales})$	-0.0911*	TFP	-0.0231
	(0.0509)		(0.0516)
Cash flow ratio	0.0285		, , , , , , , , , , , , , , , , , , ,
	(0.1536)		
Industry dummies	yes		
Country dummies	yes		
Year dummies	yes		
Ν	$276,\!801$		
Pseudo \mathbb{R}^2	0.1552		

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors in parentheses. Number of observations N. The explanatory variables are lagged one period before the deal.

	Unmatched	Me	ean		t-t	est
	Matched	Treated	Control	%bias	\mathbf{t}	p > t
Cash ratio	U	0.087	0.089	-2.0	-0.52	0.600
	М	0.087	0.083	2.7	0.53	0.599
Working capital ratio	U	0.227	0.212	6.2	1.73	0.083
	Μ	0.227	0.226	0.4	0.08	0.934
$\ln(\text{Capital})$	U	7.485	6.560	53.8	14.05	0.000
	Μ	7.485	7.484	0.0	0.00	0.996
$\ln(\text{Labour})$	U	4.487	3.514	82.0	22.57	0.000
	Μ	4.487	4.452	2.9	0.55	0.582
$\ln(\text{Labour})^2$	U	21.578	13.711	77.3	23.66	0.000
	Μ	21.578	21.207	3.7	0.65	0.517
$\Delta \ln(\text{Sales})$	U	0.019	0.023	-1.5	-0.39	0.694
	Μ	0.019	0.025	-2.1	-0.41	0.682
Cash flow ratio	U	0.091	0.084	6.0	1.92	0.055
	Μ	0.091	0.093	-1.5	-0.29	0.774
Leverage ratio	U	0.525	0.557	-14.4	-3.86	0.000
	Μ	0.525	0.518	2.7	0.52	0.600
$\ln(Wage)$	U	3.304	3.203	12.5	3.61	0.000
	Μ	3.304	3.321	-2.2	-0.41	0.685
Public	U	0.473	0.304	35.1	9.92	0.000
	Μ	0.473	0.471	0.3	0.05	0.958
$\ln(Age)$	U	2.934	2.836	12.2	3.44	0.001
	Μ	2.934	2.951	-2.1	-0.40	0.690
Intangible assets share	U	0.098	0.071	15.8	4.74	0.000
	Μ	0.098	0.109	-6.6	-1.10	0.271
TFP	U	0.116	0.058	13.4	3.75	0.000
	М	0.116	0.129	-3.0	-0.57	0.569
Propensity score	U	0.017	0.003	84.0	61.12	0.000
-	М	0.017	0.017	-0.0	-0.00	0.998

 Table 3: Propensity score matching. Testing the balancing property.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Cash ratio	0.000	-0.010^{**}	-0.012^{**}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.009	0.010	0.017^{*}
	(0.007)	(0.008)	(0.010)
$\ln(\text{Capital})$	-0.007	0.037	0.087^{**}
	(0.021)	(0.030)	(0.037)
Intangible assets ratio	0.005^{***}	0.009^{***}	0.012^{***}
	(0.002)	(0.002)	(0.003)
$\ln(\text{Labour})$	0.002	0.017	0.037^{***}
	(0.008)	(0.011)	(0.013)
$\ln(\text{Sales})$	-0.034^{**}	0.025	0.058^{***}
	(0.015)	(0.018)	(0.021)
Ν	1472	1472	1472

Table 4: Difference-in-Difference estimation for the effect of M&As on the targets' performance.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N.

Table 5: Difference-in-Difference estimation for the effect of M&As on the targets' performance,by size.

Dependent variable	$y_t -$	y_{t-1}	$y_{t+1} -$	y_{t-1}	y_{t+2} -	y_{t-1}
Explanatory variables	MA	MA x small	MA	MA x small	MA	MA x small
Cash ratio	0.006	-0.019^{*}	-0.001	-0.029^{**}	-0.004	-0.026*
	(0.005)	(0.011)	(0.006)	(0.012)	(0.006)	(0.014)
Leverage ratio	-0.010	0.002	0.000	0.032^{*}	0.008	0.029
	(0.009)	(0.014)	(0.010)	(0.018)	(0.013)	(0.022)
ln(Capital)	-0.035	0.085^{*}	-0.019	0.167^{**}	0.026	0.178**
	(0.024)	(0.047)	(0.034)	(0.069)	(0.042)	(0.083)
Intangible assets ratio	0.005^{**}	0.001	0.009^{***}	0.001	0.012^{***}	-0.001
	(0.002)	(0.004)	(0.003)	(0.005)	(0.003)	(0.006)
ln(Labour)	-0.006	0.025	-0.002	0.060**	0.010	0.083***
	(0.009)	(0.016)	(0.013)	(0.024)	(0.016)	(0.028)
$\ln(\text{Sales})$	-0.022	-0.036	0.022	0.010	0.046^{*}	0.035
	(0.017)	(0.034)	(0.022)	(0.040)	(0.025)	(0.048)
Ν	14	172	147	2	14	72

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. The regressions also include the binary variable *small* (not presented).

Deal type		Cross-border			Domestic	
Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Cash ratio	0.005	-0.011	-0.004	-0.003	-0.019^{***}	-0.021^{***}
	(0.007)	(0.008)	(0.009)	(0.006)	(0.007)	(0.008)
Leverage ratio	-0.031^{***}	-0.030^{**}	-0.011	0.012	0.041***	0.045***
	(0.011)	(0.014)	(0.017)	(0.009)	(0.011)	(0.014)
ln(Capital)	-0.034	0.031	0.033	-0.002	0.047	0.066
	(0.030)	(0.043)	(0.053)	(0.026)	(0.040)	(0.046)
Intangible assets ratio	-0.001	-0.002	-0.003	0.007^{***}	0.011^{***}	0.014^{***}
	(0.003)	(0.004)	(0.005)	(0.002)	(0.003)	(0.003)
$\ln(\text{Labour})$	0.004	0.011	0.024	0.003	0.017	0.024
	(0.011)	(0.016)	(0.021)	(0.009)	(0.015)	(0.018)
$\ln(\text{Sales})$	-0.039	0.000	0.061^{*}	-0.064^{***}	-0.007	0.004
	(0.026)	(0.031)	(0.033)	(0.019)	(0.023)	(0.027)
Ν	606	606	606	826	826	826

Table 6: Difference-in-Difference estimation for the effect of M&As on the targets' performance.Cross-border and domestic deals.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N.

Time of $M\&A$		Narrow crisis pe	eriod		Broad crisis per	iod		Non-crisis period	
Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2}-y_{t-1}$	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2}-y_{t-1}$	y_t-y_{t-1}	$y_{t+1}-y_{t-1}$	$y_{t+2}-y_{t-1}$
Cash ratio	0.001	-0.024^{*}	-0.031^{**}	-0.009	-0.024^{**}	-0.024^{**}	-0.001	-0.011*	-0.010
	(0.010)	(0.014)	(0.015)	(0.010)	(0.011)	(0.011)	(0.005)	(0.006)	(0.007)
Leverage ratio	0.000	0.021	0.033	-0.008	0.024	0.049^{**}	-0.002	0.007	0.021^{*}
1	(0.018)	(0.023)	(0.026)	(0.017)	(0.019)	(0.021)	(0.008)	(0.00)	(0.012)
$\ln(Capital)$	0.083	0.140*	0.234^{***}	-0.032	0.043	0.068	-0.053^{**}	-0.029	-0.022
	(0.061)	(0.078)	(060.0)	(0.047)	(0.059)	(0.072)	(0.024)	(0.035)	(0.043)
Intangible assets ratio	0.007	0.006	0.00	0.004	0.005	0.002	0.005^{**}	0.009^{***}	0.011^{***}
I	(0.005)	(0.006)	(0.008)	(0.005)	(0.006)	(0.00)	(0.002)	(0.003)	(0.003)
$\ln(Labour)$	0.025	0.062^{**}	0.088^{***}	-0.006	0.010	0.038	-0.008	-0.005	0.005
	(0.019)	(0.027)	(0.030)	(0.015)	(0.022)	(0.028)	(0.008)	(0.012)	(0.015)
$\ln(\mathrm{Sales})$	0.013	0.066	0.094^{*}	-0.063^{*}	0.046	0.058	-0.054^{***}	-0.012	0.015
	(0.041)	(0.045)	(0.049)	(0.032)	(0.039)	(0.043)	(0.016)	(0.019)	(0.023)
Ν	238	238	238	356	356	356	1116	1116	1116
NOTES: * $p < 0.1$, * 09/2008 - 12/2009 , E	* $p < 0.05$, storage strugglering p	*** $p < 0.01$. R eriod: 07/2007 -	obust standard e - 12/2009.	rrors in pare	ntheses. Acquis	sition in period t	. Number of obs	servations N . Nar	row crisis period:

	TISIS.
	by O
c	pertormance,
	targets
-	the
	0 D
•	ZAS
	418
	0 T
ع	tect (
	e
5	th th
¢	for
•	ation
•	estima
8	Difference
	Ľ.
Ę	Difference-1
I	
	Lable

	(1)	(2)	(3)	(4)	(5)
Panel A: Investment					
CF	0.089^{***} (0.012)	0.083^{***} (0.012)	0.082^{***} (0.012)	0.082^{***} (0.013)	0.078^{***} (0.015)
after	0.050^{**} (0.021)	0.051^{**} (0.021)	0.044^{**} (0.022)	0.050** (0.023)	0.063^{***} (0.022)
after x CF	-0.025^{*} (0.014)	-0.024 (0.015)	-0.026^{*} (0.014)	-0.023 (0.015)	-0.039^{**} (0.019)
$\Delta \ln(\text{Sales})$	``´´	0.149^{***} (0.031)	0.129^{***} (0.031)	0.149*** (0.032)	× /
$\ln(\text{Total assets})$			0.159^{***} (0.024)	0.146^{***} (0.025)	
Deal x CF					$0.029 \\ (0.025)$
Ν	11570	11462	11462	10781	11570
F-test	0.000	0.000	0.000	0.000	0.000
Panel B: $\triangle Cash$					
CF	0.202^{***} (0.016)	0.191^{***} (0.016)	0.191^{***} (0.016)	0.184^{***} (0.016)	0.206^{***} (0.021)
after	0.006^{**} (0.003)	0.006^{**} (0.003)	0.006^{**} (0.003)	0.003 (0.003)	0.006 [*] (0.003)
after x CF	-0.113^{***} (0.027)	-0.112^{***} (0.026)	-0.116^{***} (0.026)	-0.118^{***} (0.030)	-0.109^{***} (0.032)
$\Delta \ln(\text{Sales})$		0.016^{***} (0.004)	0.015^{***} (0.004)	0.015^{***} (0.004)	
$\ln(\text{Total assets})$			0.012^{***} (0.003)	0.012^{***} (0.003)	
Deal x CF					-0.008 (0.032)
Ν	11545	11437	11436	10747	11545
F-test	0.000	0.000	0.001	0.008	0.000

Table 8: The effect of M&As on the investment-cash flow and cash-cash flow sensitivity.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors (clustered at the firm level) in parentheses. Number of observations N. Investment = Investment in capital stock, $\triangle Cash$ = Change in cash holdings between two consecutive years, CF = Cash flow, after = 1 after acquisition. The regressions include year dummies. (1) presents the results of a fixed effects regression using all acquired and matched control firms, (2) adds sales growth, (3) additionally adds the log of total assets as a proxy for size, (4) drops the year of the deal for acquired firms, and (5) adds a dummy Deal (=1 for acquired firms, =0 for non-acquired firms). *F-test*: Tests null hypothesis that the cash flow sensitivity is zero after acquisition (p-values are reported).

	Dep	e: TFP		
Sample	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$	Ν
All	-0.005	0.017	0.019	1472
	(0.010)	(0.011)	(0.012)	
Cross-border	-0.011	-0.002	0.019	606
	(0.015)	(0.016)	(0.022)	
Domestic	-0.020^{*}	0.005	-0.005	826
	(0.011)	(0.014)	(0.014)	
Crisis	-0.004	0.018	0.000	356
	(0.019)	(0.021)	(0.023)	
Non-crisis	-0.012	0.007	0.007	1116
	(0.009)	(0.012)	(0.015)	
Low productive target	0.030	0.049**	0.052**	1472
	(0.020)	(0.020)	(0.023)	
Medium and high productive target	-0.020^{*}	0.003	0.004	1472
	(0.012)	(0.013)	(0.014)	

Table 9: Difference-in-Difference estimation for the effect of M&As on the targets' productivity.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. Sample covers the respective acquired firms and their matched control firms. Crisis describes the broad crisis period (07/2007 - 12/2009). In the last set of regressions, the presented coefficients for Low productive target (respectively Medium and high productive target) represent $\hat{\beta}_{1,DID} + \hat{\beta}_{2,DID}$ (respectively $\hat{\beta}_{1,DID}$), see Equation 5 when replacing small with lowprod.

A Appendix

Country	Country Code	Frequency	Percent
Austria	AT	2	0.27
Bosnia and Herzegovina	BA	3	0.41
Belgium	BE	46	6.25
Bulgaria	BG	11	1.49
Czech Republic	CZ	39	5.30
Germany	DE	41	5.57
Estonia	\mathbf{EE}	4	0.54
Spain	\mathbf{ES}	94	12.77
Finland	$_{ m FI}$	48	6.52
France	FR	126	17.12
Croatia	$_{\mathrm{HR}}$	12	1.63
Hungary	HU	15	2.04
Italy	IT	116	15.76
Norway	NO	3	0.41
Poland	PL	11	1.49
Portugal	\mathbf{PT}	15	2.04
Romania	RO	2	0.27
Serbia	\mathbf{RS}	15	2.04
Sweden	\mathbf{SE}	92	12.50
Slovenia	\mathbf{SI}	6	0.82
Slovakia	\mathbf{SK}	4	0.54
Ukraine	$\mathbf{U}\mathbf{A}$	31	4.21
Total		736	100.00

 Table A.1: Distribution of target countries.

 Table A.2:
 Variable definitions.

Variable	Definition
Total assets	Fixed assets $+$ Current assets (TOAS)
Fixed assets	Total amount (after depreciation) of non-current assets (FIAS)
	(Intangible assets + Tangible assets + Other fixed assets)
Capital (stock)	Tangible fixed assets (TFAS)
Intangible assets	Intangible fixed assets (IFAS)
Depreciation	Total amount of depreciation and amortization of the assets (DEPRE)
Investment	Investment in tangible fixed assets
	(Capital - L1.Capital + Depreciation*(Capital/Fixed assets))
Cash	Cash and cash equivalents (CASH)
$\triangle Cash$	Cash - L1.Cash
Cash flow	Cash flow (CF)
Current liabilities	Current liabilities of the companys (loans + creditors + other current liabilities) (CULI)
Working capital	Current assets - Current liabilities
Labour	Total number of employees (EMPL)
Sales	Total operating revenues (Net sales + other operating revenues + stock variations) (OPRE)
\triangle Sales	Sales - L1.Sales
Long-term debt	Long-term financial debts (e.g. to credit institutions (loans and credits), bonds) (LTDB)
Leverage ratio	(Long-term debt + Current liabilities) / Total assets
Wage	Costs of employees (STAF) / Labour
TFP	Total factor productivity (deviation from industry mean); own calculation
Age	Actual year - Year of incorporation
Public	=1 if public limited company $=0$ otherwise

NOTES: The variables are measured annually. Source: Amadeus.

Table A.3: Difference-in-Difference estimation for the effect of M&As on the targets' performance.Additional variables.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Long-term debt ratio	0.010*	0.017^{**}	0.021***
	(0.005)	(0.007)	(0.008)
Current Liabilities ratio	-0.019^{***}	-0.007	-0.004
	(0.007)	(0.008)	(0.010)
$\ln(\text{Intangible assets})$	0.110^{*}	0.301***	0.441^{***}
	(0.059)	(0.083)	(0.099)
Ν	1472	1472	1472

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. The long-term debt ratio (current liabilities ratio) is defined as long-term debt (current liabilities) over total assets.

	ln(Inta	ngible as	ssets $+1$)	Intang	ible asse	ts ratio
	Mean	SD	T-test	Mean	SD	T-test
t-1	3.610	3.212		0.021	0.063	
t	3.834	3.204	0.025	0.020	0.056	0.753
t+1	3.922	3.197	0.021	0.020	0.054	0.847
t+2	3.984	3.218	0.008	0.019	0.050	0.586
Ν	201			182		

 $\label{eq:table A.4: Change in acquirers' characteristics after a deal.$

NOTES: Acquisition in period t. T-test reports the pvalue of a T-test on equality of mean values in t (resp. t+1, t+2) and t-1. Number of observations N.

	Cross-border	Domestic
Cash ratio	0.1599	0.0108
	(0.1940)	(0.1620)
Working capital ratio	0.0218	0.3805^{***}
	(0.1233)	(0.1093)
ln(Capital)	0.0536**	0.0325
	(0.0214)	(0.0188)
ln(Labour)	0.4163***	0.5289***
× /	(0.1073)	(0.0924)
$\ln(\text{Labour})^2$	-0.0207^{*}	-0.0389^{***}
	(0.0112)	(0.0103)
$\Delta \ln(\text{Sales})$	-0.0504	-0.1076
	(0.0661)	(0.0686)
Cash flow ratio	-0.2287	0.2560
	(0.2263)	(0.1882)
Leverage ratio	0.1117	0.2229**
	(0.1225)	(0.1030)
$\ln(Wage)$	0.2568^{***}	0.1940^{***}
	(0.0722)	(0.0682)
Public	0.0151	0.0797^{*}
	(0.0480)	(0.0426)
$\ln(Age)$	-0.0564^{**}	0.0220
	(0.0255)	(0.0243)
Intangible assets share	0.5736^{***}	0.4066^{***}
	(0.1166)	(0.1144)
TFP	0.1117^{*}	-0.1436^{**}
	(0.0651)	(0.0722)
Industry dummies	yes	yes
Country dummies	yes	yes
Year dummies	yes	yes
Ν	275,729	$276,\!337$
Pseudo R^2	0.1504	0.1652

Table A.5: Probit regressions. Prediction of cross-border and domestic M&A deals.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors in parentheses. Number of observations N. The explanatory variables are lagged one period before the deal.

	Unmatched	Me	ean		t-t	est
	Matched	Treated	Control	%bias	t	p > t
Panel A: Cross-border deals						
Cash ratio	U	0.079	0.089	-8.8	-1.44	0.150
	M	0.079	0.075	3.1	0.41	0.684
Working capital ratio	U	0.201	0.212	-4.5	-0.80	0.422
	Μ	0.201	0.190	4.3	0.53	0.599
$\ln(\text{Capital})$	U	7.856	6.558	76.6	12.67	0.000
ln (Labour)	M	7.856	7.911	-3.2	-0.41	0.679
III(Labour)	U M	4.707	5.014 4 756	100.4	0.11	0.000
$\ln(\text{Labour})^2$	U	24.127	13.712	99.5	20.11	0.000
m(Lassar)	M	24.127	23.793	3.2	0.37	0.710
$\Delta \ln(\text{Sales})$	U	0.030	0.023	2.0	0.37	0.713
	Μ	0.030	0.033	-0.9	-0.10	0.919
Cash flow ratio	U	0.085	0.084	0.7	0.14	0.889
-	M	0.085	0.086	-1.0	-0.11	0.910
Leverage ratio	U	0.536	0.557	-9.1	-1.58	0.113
	M	0.536	0.530	3.0	0.37	0.713
In(wage)	U M	3.341	3.203	17.0	3.20 0.15	0.001
Public	U	0.455	0.304	31.5	5.72	0.000
1 done	M	0.455	0.465	-2.1	-0.24	0.807
$\ln(Age)$	U	2.847	2.835	1.5	0.27	0.789
	Μ	2.847	2.823	2.9	0.34	0.733
Intangible assets share	U	0.112	0.071	23.6	4.66	0.000
	Μ	0.112	0.086	15.2	1.72	0.085
TFP	U	0.154	0.058	22.6	4.00	0.000
Duon angitar agana	M	0.154	0.147	1.8	0.20	0.844
Propensity score	M	0.008	0.001	-0.0	40.39	0.000
	101	0.000	0.000	0.0	0.00	0.000
Panel B: Domestic deals						
Cash ratio	U	0.093	0.089	3.7	0.76	0.448
XX7 1 · · · · · · · · · · · · · · · · · ·	M	0.093	0.087	4.9	0.71	0.475
working capital ratio	U	0.244 0.244	0.212 0.235	13.1	2.73	0.000
ln(Capital)	U	0.244 7 232	6.250	39.2	7.67	0.098
m(cupital)	M	7.232	7.175	3.3	0.49	0.627
$\ln(\text{Labour})$	U	4.255	3.513	63.6	12.91	0.000
	Μ	4.255	4.232	2.0	0.29	0.772
$\ln(\text{Labour})^2$	U	19.464	13.706	59.4	12.99	0.000
	M	19.464	19.265	2.0	0.28	0.781
$\Delta \ln(\text{Sales})$	U	0.013	0.023	-3.3	-0.64	0.520
Cash flow ratio		0.013	0.002	3.4 10.3	0.57	0.570
Cash how fatto	M	0.090	0.084	10.5	2.04	0.008
Leverage ratio	U	0.000 0.521	0.050 0.557	-15.9	-3.18	0.001
	Μ	0.521	0.509	5.7	0.85	0.396
$\ln(Wage)$	U	3.336	3.203	17.0	3.56	0.000
	Μ	3.336	3.281	7.1	0.99	0.321
Public	U	0.477	0.304	36.0	7.63	0.000
	M	0.477	0.499	-4.5	-0.63	0.532
in(Age)	U	3.012	2.836	22.2	4.63	0.000
Intangible assets chare	I I IVI	0.012 0.086	2.940 0.071	8.3 0.2	1.22	0.224 0.047
intaligible assets shale	M	0.080	0.071	9.4 3.4	1.99 0.46	0.644
TFP	U	0.110	0.000	12.3	2.53	0.012
	M	0.110	0.080	7.1	1.06	0.290
Propensity score	U	0.012	0.001	82.3	52.32	0.000
	Μ	0.012	0.012	0.0	0.00	0.999

 Table A.6: Testing the balancing property after matching. Cross-border and domestic M&A deals.

Cash ratio	0.3694	Leverage ratio	-0.3251
	(0.5939)		(0.3885)
Working capital ratio	-0.9843^{***}	$\ln(Wage)$	0.3789
	(0.3703)		(0.2458)
$\ln(\text{Capital})$	0.0677	Public	-0.2103
	(0.0665)		(0.1370)
$\ln(\text{Labour})$	0.2673	$\ln(Age)$	-0.1259*
	(0.3250)		(0.0721)
$\ln(\text{Labour})^2$	-0.0092	Intangible assets share	0.1601
	(0.0337)		(0.3562)
$\Delta \ln(\text{Sales})$	0.1036	TFP	0.6802^{***}
	(0.1827)		(0.2565)
Cash flow ratio	-0.4777		
	(0.6060)		
Industry dummies	yes		
Country dummies	yes		
Year dummies	yes		
Ν	714		
Pseudo \mathbb{R}^2	0.2412		

 Table A.7: Probit regression. Prediction of cross-border M&A deals. Domestic deals as control group.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors in parentheses. Number of observations N. The explanatory variables are lagged one period before the deal.

	Unmatched	Me	ean		t-t	est
	Matched	Treated	Control	%bias	t	p > t
Cash ratio	U	0.078	0.094	-13.2	-1.73	0.085
	Μ	0.078	0.074	4.0	0.52	0.601
Working capital ratio	U	0.199	0.244	-18.0	-2.38	0.018
	Μ	0.201	0.201	0.3	0.03	0.976
ln(Capital)	U	7.851	7.226	38.6	5.08	0.000
	Μ	7.761	7.724	2.3	0.26	0.794
ln(Labour)	U	4.768	4.252	43.8	5.80	0.000
	Μ	4.697	4.798	-8.6	-0.98	0.328
$\ln(\text{Labour})^2$	U	24.143	19.432	42.6	5.68	0.000
	Μ	23.406	24.540	-10.2	-1.14	0.257
$\Delta \ln(\text{Sales})$	U	0.030	0.015	4.7	0.63	0.531
	Μ	0.028	0.035	-2.0	-0.20	0.838
Cash flow ratio	U	0.085	0.097	-10.0	-1.30	0.194
	Μ	0.084	0.106	-18.0	-2.26	0.024
Leverage ratio	U	0.538	0.522	7.2	0.95	0.341
	Μ	0.540	0.525	6.7	0.08	0.423
$\ln(Wage)$	U	3.338	3.338	-0.0	-0.00	0.999
	Μ	3.337	3.277	7.4	0.86	0.388
Public	U	0.454	0.478	-4.9	-0.65	0.517
	Μ	0.474	0.452	4.4	0.52	0.607
$\ln(Age)$	U	2.846	3.019	-20.9	-2.78	0.006
	Μ	2.862	2.660	24.4	2.59	0.010
Intangible assets share	U	0.112	0.086	14.3	1.90	0.058
÷	Μ	0.113	0.133	-10.9	-1.08	0.281
TFP	U	0.151	0.111	9.5	1.25	0.211
	Μ	0.137	0.124	3.1	0.37	0.709
Propensity score	U	0.593	0.301	130.7	17.36	0.000
	Μ	0.555	0.555	0.1	0.01	0.992

 Table A.8: Propensity score matching. Testing the balancing property. Domestic deals as control group.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Cash ratio	0.012	0.011	0.008
	(0.010)	(0.011)	(0.013)
Leverage ratio	-0.043^{**}	-0.052^{**}	-0.059^{**}
	(0.017)	(0.020)	(0.025)
$\ln(\text{Capital})$	0.012	0.044	0.069
	(0.043)	(0.055)	(0.066)
Intangible assets ratio	-0.004	-0.001	-0.003
	(0.004)	(0.005)	(0.005)
$\ln(\text{Labour})$	0.010	0.036	0.054^{**}
	(0.015)	(0.024)	(0.027)
$\ln(\text{Sales})$	0.040	0.058	0.070
	(0.034)	(0.038)	(0.044)
Ν	405	405	405

Table A.9: Difference-in-Difference estimation for the effect of M&As on the targets' performance, by deal type. Domestic deals as control group.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N.

	Narrow crisis period	Broad crisis period	Non-crisis period
Cash ratio	0.2817	0.2024	0.0249
	(0.3131)	(0.2486)	(0.1531)
Working capital ratio	0.3545^{*}	0.2563	0.2282**
	(0.1993)	(0.1657)	(0.1013)
$\ln(\text{Capital})$	0.0447	0.0425	0.0424**
、 <u>-</u> ,	(0.0350)	(0.0286)	(0.0173)
$\ln(\text{Labour})$	0.2672*	0.3250**	0.5183***
× ,	(0.1602)	(0.1323)	(0.0850)
$\ln(\text{Labour})^2$	-0.0054	-0.0146	-0.0333^{***}
	(0.0170)	(0.0141)	(0.0093)
$\Delta \ln(\text{Sales})$	-0.0561	-0.0884	-0.1026^{*}
	(0.1273)	(0.1041)	(0.0590)
Cash flow ratio	-0.1492	0.1435	0.0241
	(0.3472)	(0.2732)	(0.1859)
Leverage ratio	0.3582^{*}	0.2074	0.1535
	(0.2017)	(0.1671)	(0.0990)
$\ln(Wage)$	0.2199^{*}	0.2791^{***}	0.2111***
	(0.1161)	(0.0944)	(0.0623)
Public	0.1195	0.2021^{***}	0.0272
	(0.0790)	(0.0666)	(0.0393)
$\ln(Age)$	0.0721	0.0287	-0.0380^{*}
	(0.0438)	(0.0351)	(0.0217)
Intangible assets share	0.6500^{***}	0.4748^{***}	0.5545^{***}
	(0.2031)	(0.1718)	(0.0986)
TFP	-0.0884	-0.1028	0.0032
	(0.1222)	(0.0986)	(0.0616)
Industry dummies	yes	yes	yes
Country dummies	yes	yes	yes
Year dummies	yes	yes	yes
Ν	88,171	131,240	180,394
Pseudo R^2	0.1402	0.1267	0.1719

 Table A.10:
 Probit regressions.
 Prediction of crisis and non-crisis M&A deals.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors in parentheses. Number of observations N. The explanatory variables are lagged one period before the deal.

	Unmatched	Me	ean		t-1	test
	Matched	Treated	Control	%bias	t	p > t
Panel A: Narrow crisis period						
Cash ratio	U	0.073	0.087	-11.6	-1.27	0.202
	Μ	0.073	0.076	-2.4	-0.19	0.847
Working capital ratio	U	0.215	0.209	2.5	0.27	0.78_{-}
	Μ	0.215	0.220	-2.0	-0.16	0.869
ln(Capital)	U	7.649	6.499	69.2	7.17	0.00
	Μ	7.649	7.622	1.6	0.13	0.899
ln(Labour)	U	4.674	3.495	97.2	11.00	0.00
_	Μ	4.674	4.609	5.4	0.41	0.68
$\ln(\text{Labour})^2$	U	23.419	13.579	93.1	11.91	0.00
	Μ	23.419	22.702	6.8	0.47	0.642
$\Delta \ln(\text{Sales})$	U	-0.001	0.044	-16.4	-1.70	0.08
	Μ	-0.001	0.015	-5.9	-0.49	0.62
Cash flow ratio	U	0.074	0.092	-13.1	-2.06	0.03
	Μ	0.074	0.069	3.6	0.27	0.78
Leverage ratio	U	0.543	0.566	-9.7	-1.08	0.28
	М	0.543	0.552	-3.7	-0.28	0.78
$\ln(Wage)$	U	3.044	3.183	-15.3	-1.87	0.06
	Μ	3.044	3.216	-18.9	-1.47	0.14
Public	U	0.513	0.293	45.9	5.26	0.00
	Μ	0.513	0.529	-3.5	-0.26	0.79
$\ln(Age)$	U	3.039	2.813	25.3	3.17	0.00
	Μ	3.039	3.041	-0.2	-0.02	0.98
Intangible assets share	U	0.090	0.070	11.9	1.44	0.15
	Μ	0.090	0.126	-21.0	-1.30	0.19
TFP	U	-0.001	0.066	-14.8	-1.70	0.08
	Μ	-0.001	0.046	-10.2	-0.78	0.43
Propensity score	U	0.008	0.001	72.9	25.80	0.00
	М	0.008	0.009	-1.8	-0.10	0.92
Panel B: Broad crisis period						
Cash ratio	U	0.080	0.085	-4.7	-0.65	0.51
	M	0.080	0.082	-1.6	-0.15	0.88
Working capital ratio	U	0.222	0.203	7.7	1.05	0.29
torning capital fatto	M	0.222	0.228	-2.5	-0.24	0.81
n(Capital)	U	7.563	6.528	63.4	7.89	0.00
(Capital)	M	7.563	7 455	6.6	0.65	0.51
n(Labour)	U	4.628	3.533	90.7	12.39	0.00
	й	4.628	4.543	7.0	0.68	0.49
$\ln(\text{Labour})^2$	U	22.929	13.868	87.0	13.26	0.00
	M	22.929	21.827	10.6	0.96	0.34
$\wedge \ln(\text{Sales})$	U	0.030	0.065	-12.4	-1.57	0.11
(> allos)	M	0.030	0.041	-4.0	-0.37	0.71
Cash flow ratio	Ū	0.093	0.093	0.2	0.03	0.97
	м	0.093	0 103	_7 7	-0.72	0 47
Leverage ratio	I	0.525	0.569	-19.1	-2.63	0.00
Lettinge ratio	м	0.525 0.525	0.534	-4.3	-0.39	0.00
In(Wage)	I	3.020	3155	-9.1	-1.35	0.17
(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	м	3 070	3 140	-7.5	-0.65	0.51
Public	IT	0.522	0.140	47.9	6.60	0.01
	м	0.522 0.522	0.200 0.545	_4 7	-0.42	0.67
n(Age)	II	2 960	2 792	20.6	3 01	0.00
m(1180)	M	2.909 2.060	2.192	_ & &	-0.84	0.00
Intensible essets share	TAT TAT	2.909 0.084	0.040	-0.0	-0.04	0.40
intangible assets sliale	M	0.004	0.070	0.0 _Q /	-0.60	0.24
TFP	TAT T	0.004	0.097	-0.4 79	-0.09	0.48
1 F F	U	0.024	0.008	-1.3	-1.03	0.30
Proponsity soons		0.024	0.064	-8.0 74.0	-0.82	0.41
r ropensity score	U	0.007	0.001	(4.9	20.87	0.00
	M	0.007	0.007	-0.4	-0.02	0.98

 Table A.11: Testing the balancing property after matching. Deals in the crisis.

	Unmatched	Me	ean		t-t	est
	Matched	Treated	Control	%bias	\mathbf{t}	$p\!> t $
Cash ratio	U	0.089	0.091	-1.7	-0.39	0.695
	Μ	0.089	0.091	-1.6	-0.27	0.788
Working capital ratio	U	0.229	0.215	5.4	1.31	0.189
	Μ	0.229	0.220	3.5	0.57	0.567
$\ln(\text{Capital})$	U	7.460	6.578	50.4	11.54	0.000
	Μ	7.460	7.498	-2.2	-0.36	0.719
$\ln(\text{Labour})$	U	4.442	3.512	78.8	18.80	0.000
	Μ	4.442	4.451	-0.8	-0.13	0.897
$\ln(\text{Labour})^2$	U	21.148	13.692	73.9	19.56	0.000
	Μ	21.148	21.142	0.1	0.01	0.993
$\Delta \ln(\text{Sales})$	U	0.015	0.012	1.0	0.23	0.818
	Μ	0.015	0.021	-1.7	-0.33	0.744
Cash flow ratio	U	0.090	0.080	9.9	2.61	0.009
	Μ	0.090	0.094	-3.5	-0.58	0.563
Leverage ratio	U	0.525	0.553	-12.4	-2.87	0.004
	Μ	0.525	0.518	2.9	0.49	0.622
$\ln(Wage)$	U	3.378	3.225	20.3	5.00	0.000
	Μ	3.378	3.339	5.2	0.84	0.399
Public	U	0.457	0.311	30.3	7.42	0.000
	Μ	0.457	0.455	0.4	0.06	0.952
$\ln(Age)$	U	2.922	2.853	8.8	2.12	0.034
	Μ	2.922	2.966	-5.6	-0.93	0.352
Intangible assets share	U	0.102	0.072	17.1	4.52	0.000
	Μ	0.102	0.096	3.5	0.55	0.585
TFP	U	0.145	0.060	20.2	4.87	0.000
	М	0.145	0.147	-0.4	-0.07	0.944
Propensity score	U	0.021	0.003	93.2	55.77	0.000
-	Μ	0.021	0.021	0.0	0.00	0.998

 Table A.12: Testing the balancing property after matching. Deals outside the crisis.

n the	
eals ir	
ity. D	
ntensi	
gical i	
golou	
y tech	
ace, b	
ormai	
' perf	
argets	
the t	
As on	
f M&1	
fect of	
the ef	
n for	
natio	
e estin	
erence	
n-Diff	
ence-ii	
Differ	
.13:]	
ole A	is.
Tat	crisi

Dependent variable		y_t-y_{t-1}			$y_{t+1}-y_{t-1}$			$y_{t+2}-y_{t-1}$	
Explanatory variables	MA	$MA \ge hightech$	hightech	MA	MA x hightech	hightech	\mathbf{MA}	$MA \ge hightech$	hightech
Cash ratio	0.007	-0.014	0.013	-0.009	-0.033	0.029*	-0.011	-0.045	0.042^{**}
	(0.016)	(0.020)	(0.011)	(0.018)	(0.028)	(0.016)	(0.019)	(0.030)	(0.019)
Leverage ratio	0.006	-0.015	0.015	0.025	-0.009	0.045*	0.025	0.019	0.025
	(0.024)	(0.036)	(0.019)	(0.030)	(0.047)	(0.024)	(0.034)	(0.052)	(0.028)
$\ln(Capital)$	-0.008	0.212^{*}	-0.170^{***}	-0.017	0.367^{**}	-0.294^{***}	0.063	0.397^{**}	-0.293^{***}
	(0.087)	(0.118)	(0.059)	(0.108)	(0.154)	(0.096)	(0.119)	(0.179)	(0.110)
Intangible assets ratio	0.009	-0.004	-0.003	0.006	0.001	-0.006	0.010	-0.004	-0.013
	(0.006)	(0.011)	(0.008)	(0.006)	(0.013)	(0.011)	(0.008)	(0.016)	(0.012)
$\ln(Labour)$	0.035	-0.022	-0.016	0.042	0.046	-0.044	0.065	0.053	-0.038
	(0.023)	(0.040)	(0.026)	(0.035)	(0.056)	(0.035)	(0.040)	(0.062)	(0.038)
$\ln(\mathrm{Sales})$	-0.002	0.035	-0.143^{**}	0.037	0.068	-0.035	0.087	0.017	0.077
	(0.049)	(0.085)	(0.061)	(0.055)	(0.094)	(0.063)	(0.062)	(0.100)	(0.065)
N		238			238			238	
NOTES: * $p < 0.1$, ** firms between 09/2008	p < 0.05, *	** $p < 0.01$. Robust 100 (narrow crisis m	standard errors	in parenthes	ses. Acquisition in I trol firms from the	period t . Number separate matchin	of observat	ons N. The sample by time period (se	covers acquired

Hightech = 1 if firm operates in a technology-intensive industry (NACE Rev. 2, 2-digit industry codes 20, 21, 26, 27-30). 43% of acquired firms are classified as hightech.

Dependent variable		$y_t - y_{t-1}$			$y_{t+1} - y_{t-1}$			$y_{t+2} - y_{t-1}$	
Explanatory variables	MA	MA x hightech	hightech –	MA	MA x hightech	hightech	MA	MA x hightech	hightech
Cash ratio	0.000	-0.002	-0.008	-0.012^{*}	0.005	-0.004	-0.011^{*}	-0.003	0.007
	(0.006)	(0.010)	(0.006)	(0.006)	(0.011)	(0.007)	(0.007)	(0.013)	(0.007)
Leverage ratio	-0.012	0.011	-0.006	0.001	0.029	-0.009	0.002	0.047^{*}	-0.016
	(0.009)	(0.014)	(0.001)	(0.010)	(0.019)	(0.010)	(0.012)	(0.024)	(0.013)
$\ln(\text{Capital})$	-0.015	0.023	0.030	0.014	0.072	-0.024	0.064	0.072	-0.067
	(0.026)	(0.042)	(0.026)	(0.036)	(0.065)	(0.043)	(0.043)	(0.082)	(0.057)
Intangible assets ratio	0.007^{***}	-0.004	0.000	0.010^{***}	-0.001	0.000	0.011^{***}	0.001	-0.002
	(0.002)	(0.004)	(0.002)	(0.003)	(0.005)	(0.003)	(0.003)	(0.006)	(0.003)
$\ln(Labour)$	-0.008	0.029^{*}	-0.026^{**}	0.008	0.028	-0.011	0.038^{**}	-0.004	0.018
	(0.009)	(0.016)	(0.011)	(0.014)	(0.023)	(0.016)	(0.017)	(0.027)	(0.018)
$\ln(Sales)$	-0.018	-0.049	-0.006	0.029	-0.012	0.011	0.079^{***}	-0.068	0.056^{*}
	(0.016)	(0.035)	(0.018)	(0.022)	(0.040)	(0.025)	(0.026)	(0.046)	(0.029)
Ν		1472			1472			1472	
NOTES: * $p < 0.1$, ** p in a technology-intensive	< 0.05, *** p industry (NAG	< 0.01. Robust star CE Rev. 2, 2-digit i	ndard errors in _I industry codes 20	barentheses. Act 0, 21, 26, 27-30)	quisition in period . 43% of acquired	t . Number of ϵ	observations N ified as <i>highted</i>	^{r} . Hightech = 1 if f h.	firm operates

	s.
ĺ	ea.
5	
-	A
	ensity.
	Int
-	a
•	2 GIC
-	JOIC
-	Sch
	V t
-	<u>ک</u>
·	pertormance
•	Š
	target
-	che
	n D
	1s
	\mathcal{K}_{F}
2	Σ
د	01
ε	ettect
_	he
	r
c	5
•	tioi
	ma
•	St1
	ence e
٤	ttei
:	Ē
	ш-
	G.
	ren
ε	щe
ĥ	5
,	4
۲	Ì
1	d G
-	ЪĘ
	~

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Panel A: Matching with replacement			
Cash ratio	0.000	-0.010^{*}	-0.011^{**}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.009	0.011	0.017
	(0.007)	(0.008)	(0.010)
$\ln(\text{Capital})$	-0.013	0.036	0.079^{**}
	(0.021)	(0.031)	(0.037)
Intangible assets ratio	0.005***	0.009***	0.011***
	(0.002)	(0.002)	(0.003)
ln(Labour)	(0.000)	0.016	0.034^{**}
lr(Color)	(0.008)	(0.011)	(0.013)
In(Sales)	-0.057	(0.025)	(0.032)
	(0.013)	(0.019)	(0.022)
N	1449	1449	1449
Panel B: Logit estimation			
Cash ratio	0.001	-0.009^{*}	-0.010^{*}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.003	0.017**	0.032***
0	(0.007)	(0.009)	(0.010)
ln(Capital)	0.010	0.043	0.047
	(0.019)	(0.028)	(0.035)
Intangible assets ratio	0.005***	0.008***	0.009***
	(0.002)	(0.002)	(0.003)
$\ln(\text{Labour})$	0.010	0.014	0.023^{*}
	(0.007)	(0.011)	(0.013)
$\ln(\text{Sales})$	-0.036^{**}	0.006	0.030
	(0.015)	(0.018)	(0.021)
Ν	1472	1472	1472
Panel C: Control for pre-acquisition trends			
Cash ratio	-0.001	-0.009^{*}	-0.014^{**}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.003	0.007	0.022*
	(0.007)	(0.009)	(0.011)
$\ln(\text{Capital})$	-0.008	0.001	0.005
	(0.021)	(0.031)	(0.037)
Intangible assets ratio	0.005***	0.010***	0.012***
	(0.002)	(0.002)	(0.003)
$\ln(\text{Labour})$	0.006	0.017	0.024*
. (~	(0.008)	(0.011)	(0.013)
$\ln(\text{Sales})$	-0.029^{*}	0.010	0.023
	(0.017)	(0.019)	(0.022)
Ν	1370	1370	1370

Table A.15: Robustness check: One-to-one nearest neighbor matching on the propensity score.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Panel A: Caliper matching			
Cash ratio	0.000	-0.011^{**}	-0.012^{**}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.009	0.011	0.018^{*}
	(0.007)	(0.008)	(0.010)
$\ln(\text{Capital})$	-0.011	0.038	0.088^{**}
	(0.021)	(0.030)	(0.037)
Intangible assets ratio	0.005^{***}	0.009^{***}	0.012^{***}
	(0.002)	(0.002)	(0.003)
$\ln(\text{Labour})$	0.002	0.019^{*}	0.039^{***}
	(0.008)	(0.011)	(0.013)
$\ln(\text{Sales})$	-0.033^{**}	0.027	0.061^{***}
	(0.015)	(0.018)	(0.022)
Ν	1460	1460	1460
Panel B: Radius matching			
Cash ratio	-0.002	-0.013^{***}	-0.016^{***}
	(0.004)	(0.004)	(0.005)
Leverage ratio	0.001	0.027^{***}	0.042^{***}
	(0.006)	(0.007)	(0.009)
$\ln(\text{Capital})$	-0.033^{**}	-0.016	0.000
	(0.016)	(0.023)	(0.027)
Intangible assets ratio	0.004^{**}	0.005^{***}	0.006^{***}
	(0.002)	(0.002)	(0.002)
$\ln(\text{Labour})$	-0.007	-0.008	0.000
	(0.005)	(0.008)	(0.010)
$\ln(\text{Sales})$	-0.028^{**}	0.035^{**}	0.077^{***}
	(0.013)	(0.014)	(0.016)
Ν	275,380	$275,\!380$	$275,\!380$

 Table A.16: Robustness check: Matching algorithm.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. The caliper is set to 0.0002.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Panel A: Mahalanobis matching 1			
Cash ratio	-0.003	-0.012^{**}	-0.016^{***}
	(0.005)	(0.005)	(0.006)
Leverage ratio	0.004	0.019^{**}	0.026**
	(0.007)	(0.009)	(0.011)
$\ln(\text{Capital})$	0.005	0.031	0.047
T , 111 ,	(0.020)	(0.029)	(0.035)
Intangible assets ratio	0.005^{***}	0.008^{***}	0.008^{***}
ln(Labour)	(0.002)	(0.002)	(0.003) 0.032**
m(Labour)	(0.008)	(0.010)	(0.032)
ln(Sales)	(0.003) -0.027*	0.016	0.039*
in(Seco)	(0.016)	(0.019)	(0.021)
Ν	1448	1448	1448
	1440	1448	1440
Panel B: Mahalanobis matching 2			
Cash ratio	-0.001	-0.012^{**}	-0.016^{***}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.007	0.009	0.021*
	(0.007)	(0.009)	(0.011)
ln(Capital)	-0.024	0.001	0.019
T / 11 / /	(0.021)	(0.029)	(0.035)
Intangible assets ratio	(0.005)	(0.008)	(0.008^{++++})
ln(Labour)	(0.002)	(0.002)	(0.003)
III(Labour)	(0.000)	(0.003)	(0.017)
ln(Sales)	(0.007) -0.052***	0.006	0.010
in(Seco)	(0.015)	(0.019)	(0.021)
N	1449	1449	1449
	1440	1448	1440
Panel C: Kernel matching			
Cash ratio	-0.001	-0.012^{***}	-0.016^{***}
	(0.004)	(0.004)	(0.005)
Leverage ratio	0.001	0.027***	0.042***
	(0.006)	(0.007)	(0.009)
ln(Capital)	-0.033^{**}	-0.017	-0.001
Interneible eggete notic	(0.016)	(0.023)	(0.027)
intangible assets ratio	(0.004)	(0.000)	(0.000)
ln(Labour)	(0.002) -0.007	(0.002)	(0.002) -0.002
m(Labour)	(0.005)	(0.008)	(0.002)
ln(Sales)	-0.029^{**}	0.032**	0.073***
	(0.013)	(0.014)	(0.016)
Ν	276,797	276,797	276,797

 Table A.17: Robustness check: Matching algorithm.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. Mahalanobis matching 1 (respectively Mahalanobis matching 2) puts additional weight on the firm's year and country (respectively firm's year and industry).

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Cash ratio	-0.001	-0.011^{**}	-0.012^{**}
	(0.005)	(0.005)	(0.006)
Leverage ratio	-0.009	0.011	0.017
	(0.007)	(0.008)	(0.010)
$\ln(\text{Capital})$	-0.003	0.040	0.089^{**}
	(0.021)	(0.030)	(0.037)
Intangible assets ratio	0.006^{***}	0.009^{***}	0.012^{***}
	(0.002)	(0.002)	(0.003)
$\ln(\text{Labour})$	0.001	0.017	0.037^{***}
	(0.008)	(0.011)	(0.013)
$\ln(\text{Sales})$	-0.034^{**}	0.027	0.059^{***}
	(0.015)	(0.018)	(0.022)
Ν	1464	1464	1464

 Table A.18: Robustness check: Stable Unit Treatment Value Assumption (SUTVA).

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N.

Dependent variable	$y_t - y_{t-1}$	$y_{t+1} - y_{t-1}$	$y_{t+2} - y_{t-1}$
Panel A: OLS using all observations			
Cash ratio	-0.002	-0.013^{***}	-0.016^{***}
	(0.003)	(0.004)	(0.004)
Leverage ratio	-0.005	0.013*	0.023***
-	(0.006)	(0.007)	(0.008)
$\ln(\text{Capital})$	-0.007	0.024	0.028
	(0.016)	(0.023)	(0.027)
Intangible assets ratio	0.005***	0.007^{***}	0.008***
	(0.002)	(0.002)	(0.002)
ln(Labour)	0.005	0.013^{*}	0.027^{***}
	(0.005)	(0.008)	(0.010)
$\ln(\text{Sales})$	-0.033^{***}	0.010	0.029^{*}
	(0.012)	(0.014)	(0.016)
Ν	276,801	$276,\!801$	$276,\!801$
Panel B: Propensity Score Reweighting			
Cash ratio	-0.001	-0.011^{***}	-0.014^{***}
	(0.004)	(0.004)	(0.005)
Leverage ratio	-0.003	0.017**	0.027^{***}
	(0.006)	(0.008)	(0.010)
$\ln(\text{Capital})$	-0.019	0.006	0.019
	(0.016)	(0.023)	(0.027)
Intangible assets ratio	0.005^{***}	0.007***	0.008***
	(0.002)	(0.002)	(0.002)
$\ln(\text{Labour})$	-0.001	0.003	0.018^{*}
	(0.006)	(0.009)	(0.011)
$\ln(\text{Sales})$	-0.036^{**}	0.020	0.055***
	(0.014)	(0.015)	(0.017)
Ν	276,801	276,801	276,801

Table A.19: Robustness check: Estimation strategy.

NOTES: * p < 0.1, ** p < 0.05, *** p < 0.01. Robust standard errors in parentheses. Acquisition in period t. Number of observations N. The estimations are performed using target firms and all non-acquired firms. In Panel A, the change in outcomes is regressed on the dummy MA, lagged firm characteristics X_{t-1} , and time, year and country dummies. In Panel B, the change in outcomes is regressed on the dummy MA, and propensity score reweighting is applied.

PREVIOUS DISCUSSION PAPERS

- 257 Stiebale, Joel and Wößner, Nicole, M&As, Investment and Financing Constraints, July 2017.
- 256 Wellmann, Nicolas, OTT-Messaging and Mobile Telecommunication: A Joint Market? An Empirical Approach, July 2017.
- 255 Ciani, Andrea and Imbruno, Michele, Microeconomic Mechanisms Behind Export Spillovers from FDI: Evidence from Bulgaria, June 2017.
- 254 Hunold, Matthias and Muthers, Johannes, Capacity Constraints, Price Discrimination, Inefficient Competition and Subcontracting, June 2017.
- 253 Dertwinkel-Kalt, Markus and Köster, Mats, Salient Compromises in the Newsvendor Game, June 2017.
- 252 Siekmann, Manuel, Characteristics, Causes, and Price Effects: Empirical Evidence of Intraday Edgeworth Cycles, May, 2017.
- 251 Benndorf, Volker, Moellers, Claudia and Normann, Hans-Theo, Experienced vs. Inexperienced Participants in the Lab: Do they Behave Differently?, May 2017. Forthcoming in: Journal of the Economic Science Association.
- 250 Hunold, Matthias, Backward Ownership, Uniform Pricing and Entry Deterrence, May 2017.
- 249 Kreickemeier, Udo and Wrona, Jens, Industrialisation and the Big Push in a Global Economy, May 2017.
- 248 Dertwinkel-Kalt, Markus and Köster, Mats, Local Thinking and Skewness Preferences, April 2017.
- 247 Shekhar, Shiva, Homing Choice and Platform Pricing Strategy, March 2017.
- 246 Manasakis, Constantine, Mitrokostas, Evangelos and Petrakis, Emmanuel, Strategic Corporate Social Responsibility by a Multinational Firm, March 2017.
- 245 Ciani, Andrea, Income Inequality and the Quality of Imports, March 2017.
- 244 Bonnet, Céline and Schain, Jan Philip, An Empirical Analysis of Mergers: Efficiency Gains and Impact on Consumer Prices, February 2017.
- Benndorf, Volker and Martinez-Martinez, Ismael, Perturbed Best Response Dynamics in a Hawk-Dove Game, January 2017.
 Published in: Economics Letters, 153 (2017), pp. 61-64.
- 242 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Trade and Manufacturing Jobs in Germany, January 2017. Forthcoming in: American Economic Review, Papers & Proceedings.
- 241 Borrs, Linda and Knauth, Florian, The Impact of Trade and Technology on Wage Components, December 2016.
- 240 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Selling Gasoline as a By-Product: The Impact of Market Structure on Local Prices, December 2016.

- 239 Herr, Annika and Normann, Hans-Theo, How Much Priority Bonus Should be Given to Registered Organ Donors? An Experimental Analysis, November 2016.
- 238 Steffen, Nico, Optimal Tariffs and Firm Technology Choice: An Environmental Approach, November 2016.
- 237 Behrens, Kristian, Mion, Giordano, Murata, Yasusada and Suedekum, Jens, Distorted Monopolistic Competition, November 2016.
- 236 Beckmann, Klaus, Dewenter, Ralf and Thomas, Tobias, Can News Draw Blood? The Impact of Media Coverage on the Number and Severity of Terror Attacks, November 2016. Forthcoming in: Peace Economics, Peace Science and Public Policy.
- 235 Dewenter, Ralf, Dulleck, Uwe and Thomas, Tobias, Does the 4th Estate Deliver? Towars a More Direct Measure of Political Media Bias, November 2016.
- 234 Egger, Hartmut, Kreickemeier, Udo, Moser, Christoph and Wrona, Jens, Offshoring and Job Polarisation Between Firms, November 2016.
- 233 Moellers, Claudia, Stühmeier, Torben and Wenzel, Tobias, Search Costs in Concentrated Markets An Experimental Analysis, October 2016.
- 232 Moellers, Claudia, Reputation and Foreclosure with Vertical Integration Experimental Evidence, October 2016.
- 231 Alipranti, Maria, Mitrokostas, Evangelos and Petrakis, Emmanuel, Non-comparative and Comparative Advertising in Oligopolistic Markets, October 2016. Forthcoming in: The Manchester School.
- 230 Jeitschko, Thomas D., Liu, Ting and Wang, Tao, Information Acquisition, Signaling and Learning in Duopoly, October 2016.
- 229 Stiebale, Joel and Vencappa, Dev, Acquisitions, Markups, Efficiency, and Product Quality: Evidende from India, October 2016.
- 228 Dewenter, Ralf and Heimeshoff, Ulrich, Predicting Advertising Volumes: A Structural Time Series Approach, October 2016.
- 227 Wagner, Valentin, Seeking Risk or Answering Smart? Framing in Elementary Schools, October 2016.
- 226 Moellers, Claudia, Normann, Hans-Theo and Snyder, Christopher M., Communication in Vertical Markets: Experimental Evidence, July 2016. Published in: International Journal of Industrial Organization, 50 (2017), pp. 214-258.
- 225 Argentesi, Elena, Buccirossi, Paolo, Cervone, Roberto, Duso, Tomaso and Marrazzo, Alessia, The Effect of Retail Mergers on Prices and Variety: An Ex-post Evaluation, June 2016.
- 224 Aghadadashli, Hamid, Dertwinkel-Kalt, Markus and Wey, Christian, The Nash Bargaining Solution in Vertical Relations With Linear Input Prices, June 2016. Published in: Economics Letters, 145 (2016), pp. 291-294.
- 223 Fan, Ying, Kühn, Kai-Uwe and Lafontaine, Francine, Financial Constraints and Moral Hazard: The Case of Franchising, June 2016. Forthcoming in: Journal of Political Economy.

- 222 Benndorf, Volker, Martinez-Martinez, Ismael and Normann, Hans-Theo, Equilibrium Selection with Coupled Populations in Hawk-Dove Games: Theory and Experiment in Continuous Time, June 2016. Published in: Journal of Economic Theory, 165 (2016), pp. 472-486.
- 221 Lange, Mirjam R. J. and Saric, Amela, Substitution between Fixed, Mobile, and Voice over IP Telephony – Evidence from the European Union, May 2016. Published in: Telecommunications Policy, 40 (2016), pp. 1007-1019.
- 220 Dewenter, Ralf, Heimeshoff, Ulrich and Lüth, Hendrik, The Impact of the Market Transparency Unit for Fuels on Gasoline Prices in Germany, May 2016. Published in: Applied Economics Letters, 24 (2017), pp. 302-305.
- 219 Schain, Jan Philip and Stiebale, Joel, Innovation, Institutional Ownership, and Financial Constraints, April 2016.
- 218 Haucap, Justus and Stiebale, Joel, How Mergers Affect Innovation: Theory and Evidence from the Pharmaceutical Industry, April 2016.
- 217 Dertwinkel-Kalt, Markus and Wey, Christian, Evidence Production in Merger Control: The Role of Remedies, March 2016.
- 216 Dertwinkel-Kalt, Markus, Köhler, Katrin, Lange, Mirjam R. J. and Wenzel, Tobias, Demand Shifts Due to Salience Effects: Experimental Evidence, March 2016. Published in: Journal of the European Economic Association, 15 (2017), pp. 626-653.
- Dewenter, Ralf, Heimeshoff, Ulrich and Thomas, Tobias, Media Coverage and Car Manufacturers' Sales, March 2016.
 Published in: Economics Bulletin, 36 (2016), pp. 976-982.
- 214 Dertwinkel-Kalt, Markus and Riener, Gerhard, A First Test of Focusing Theory, February 2016.
- 213 Heinz, Matthias, Normann, Hans-Theo and Rau, Holger A., How Competitiveness May Cause a Gender Wage Gap: Experimental Evidence, February 2016. Forthcoming in: European Economic Review, 90 (2016), pp. 336-349.
- 212 Fudickar, Roman, Hottenrott, Hanna and Lawson, Cornelia, What's the Price of Consulting? Effects of Public and Private Sector Consulting on Academic Research, February 2016.
- 211 Stühmeier, Torben, Competition and Corporate Control in Partial Ownership Acquisitions, February 2016. Published in: Journal of Industry, Competition and Trade, 16 (2016), pp. 297-308.
- 210 Muck, Johannes, Tariff-Mediated Network Effects with Incompletely Informed Consumers, January 2016.
- 209 Dertwinkel-Kalt, Markus and Wey, Christian, Structural Remedies as a Signalling Device, January 2016.
 Published in: Information Economics and Policy, 35 (2016), pp. 1-6.
- 208 Herr, Annika and Hottenrott, Hanna, Higher Prices, Higher Quality? Evidence From German Nursing Homes, January 2016. Published in: Health Policy, 120 (2016), pp. 179-189.
- 207 Gaudin, Germain and Mantzari, Despoina, Margin Squeeze: An Above-Cost Predatory Pricing Approach, January 2016. Published in: Journal of Competition Law & Economics, 12 (2016), pp. 151-179.

- 206 Hottenrott, Hanna, Rexhäuser, Sascha and Veugelers, Reinhilde, Organisational Change and the Productivity Effects of Green Technology Adoption, January 2016. Published in: Energy and Ressource Economics, 43 (2016), pp. 172–194.
- 205 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Adjusting to Globalization – Evidence from Worker-Establishment Matches in Germany, January 2016.
- 204 Banerjee, Debosree, Ibañez, Marcela, Riener, Gerhard and Wollni, Meike, Volunteering to Take on Power: Experimental Evidence from Matrilineal and Patriarchal Societies in India, November 2015.
- 203 Wagner, Valentin and Riener, Gerhard, Peers or Parents? On Non-Monetary Incentives in Schools, November 2015.
- 202 Gaudin, Germain, Pass-Through, Vertical Contracts, and Bargains, November 2015. Published in: Economics Letters, 139 (2016), pp. 1-4.
- 201 Demeulemeester, Sarah and Hottenrott, Hanna, R&D Subsidies and Firms' Cost of Debt, November 2015.
- 200 Kreickemeier, Udo and Wrona, Jens, Two-Way Migration Between Similar Countries, October 2015. Forthcoming in: World Economy.
- Haucap, Justus and Stühmeier, Torben, Competition and Antitrust in Internet Markets, October 2015.
 Published in: Bauer, J. and M. Latzer (Eds.), Handbook on the Economics of the Internet, Edward Elgar: Cheltenham 2016, pp. 183-210.
- 198 Alipranti, Maria, Milliou, Chrysovalantou and Petrakis, Emmanuel, On Vertical Relations and the Timing of Technology, October 2015. Published in: Journal of Economic Behavior and Organization, 120 (2015), pp. 117-129.
- 197 Kellner, Christian, Reinstein, David and Riener, Gerhard, Stochastic Income and Conditional Generosity, October 2015.
- 196 Chlaß, Nadine and Riener, Gerhard, Lying, Spying, Sabotaging: Procedures and Consequences, September 2015.
- 195 Gaudin, Germain, Vertical Bargaining and Retail Competition: What Drives Countervailing Power?, May 2017 (First Version September 2015). Forthcoming in: The Economic Journal.
- 194 Baumann, Florian and Friehe, Tim, Learning-by-Doing in Torts: Liability and Information About Accident Technology, September 2015.
- 193 Defever, Fabrice, Fischer, Christian and Suedekum, Jens, Relational Contracts and Supplier Turnover in the Global Economy, August 2015. Published in: Journal of International Economics, 103 (2016), pp. 147-165.
- 192 Gu, Yiquan and Wenzel, Tobias, Putting on a Tight Leash and Levelling Playing Field: An Experiment in Strategic Obfuscation and Consumer Protection, July 2015. Published in: International Journal of Industrial Organization, 42 (2015), pp. 120-128.
- 191 Ciani, Andrea and Bartoli, Francesca, Export Quality Upgrading under Credit Constraints, July 2015.
- 190 Hasnas, Irina and Wey, Christian, Full Versus Partial Collusion among Brands and Private Label Producers, July 2015.

- 189 Dertwinkel-Kalt, Markus and Köster, Mats, Violations of First-Order Stochastic Dominance as Salience Effects, June 2015. Published in: Journal of Behavioral and Experimental Economics, 59 (2015), pp. 42-46.
- 188 Kholodilin, Konstantin, Kolmer, Christian, Thomas, Tobias and Ulbricht, Dirk, Asymmetric Perceptions of the Economy: Media, Firms, Consumers, and Experts, June 2015.
- 187 Dertwinkel-Kalt, Markus and Wey, Christian, Merger Remedies in Oligopoly under a Consumer Welfare Standard, June 2015.
 Published in: Journal of Law, Economics, & Organization, 32 (2016), pp. 150-179.
- 186 Dertwinkel-Kalt, Markus, Salience and Health Campaigns, May 2015. Published in: Forum for Health Economics & Policy, 19 (2016), pp. 1-22.
- 185 Wrona, Jens, Border Effects without Borders: What Divides Japan's Internal Trade? May 2015.
- Amess, Kevin, Stiebale, Joel and Wright, Mike, The Impact of Private Equity on Firms' Innovation Activity, April 2015.
 Published in: European Economic Review, 86 (2016), pp. 147-160.
- 183 Ibañez, Marcela, Rai, Ashok and Riener, Gerhard, Sorting Through Affirmative Action: Three Field Experiments in Colombia, April 2015.
- 182 Baumann, Florian, Friehe, Tim and Rasch, Alexander, The Influence of Product Liability on Vertical Product Differentiation, April 2015. Published in: Economics Letters, 147 (2016), pp. 55-58 under the title "Why Product Liability May Lower Product Safety".
- 181 Baumann, Florian and Friehe, Tim, Proof beyond a Reasonable Doubt: Laboratory Evidence, March 2015.
- 180 Rasch, Alexander and Waibel, Christian, What Drives Fraud in a Credence Goods Market? – Evidence from a Field Study, March 2015.
- 179 Jeitschko, Thomas D., Incongruities of Real and Intellectual Property: Economic Concerns in Patent Policy and Practice, February 2015. Forthcoming in: Michigan State Law Review.
- 178 Buchwald, Achim and Hottenrott, Hanna, Women on the Board and Executive Duration Evidence for European Listed Firms, February 2015.
- 177 Heblich, Stephan, Lameli, Alfred and Riener, Gerhard, Regional Accents on Individual Economic Behavior: A Lab Experiment on Linguistic Performance, Cognitive Ratings and Economic Decisions, February 2015. Published in: PLoS ONE, 10 (2015), e0113475.
- 176 Herr, Annika, Nguyen, Thu-Van and Schmitz, Hendrik, Does Quality Disclosure Improve Quality? Responses to the Introduction of Nursing Home Report Cards in Germany, February 2015. Published in: Health Policy, 120 (2016), pp.1162-1170.
- Herr, Annika and Normann, Hans-Theo, Organ Donation in the Lab: Preferences and Votes on the Priority Rule, February 2015.
 Published in: Journal of Economic Behavior and Organization, 131 Part B (2016), pp. 139-149.
- 174 Buchwald, Achim, Competition, Outside Directors and Executive Turnover: Implications for Corporate Governance in the EU, February 2015.

- 173 Buchwald, Achim and Thorwarth, Susanne, Outside Directors on the Board, Competition and Innovation, February 2015.
- 172 Dewenter, Ralf and Giessing, Leonie, The Effects of Elite Sports Participation on Later Job Success, February 2015.
- 171 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Price Dispersion and Station Heterogeneity on German Retail Gasoline Markets, January 2015. Forthcoming in: The Energy Journal.
- 170 Schweinberger, Albert G. and Suedekum, Jens, De-Industrialisation and Entrepreneurship under Monopolistic Competition, January 2015. Published in: Oxford Economic Papers, 67 (2015), pp. 1174-1185.

Older discussion papers can be found online at: <u>http://ideas.repec.org/s/zbw/dicedp.html</u>

Heinrich-Heine-University of Düsseldorf

Düsseldorf Institute for Competition Economics (DICE)

Universitätsstraße 1_40225 Düsseldorf www.dice.hhu.de